

NOTICE OF ANNUAL AND SPECIAL MEETING MANAGEMENT INFORMATION CIRCULAR

FOR THE

ANNUAL AND SPECIAL MEETING OF SHAREHOLDERS

TO BE HELD

FRIDAY, FEBRUARY 17, 2023 10:00 A.M. (PACIFIC) SUITE 1305, 1090 WEST GEORGIA STREET VANCOUVER, BRITISH COLUMBIA

COVID-19

In light of ongoing concerns related to the spread of COVID-19, and in order to mitigate the potential risks to the health and safety associated with COVID-19, shareholders are strongly encouraged to vote on the matters before the Meeting by proxy rather than attend the meeting in person. Shareholders wishing to attend the Meeting in person must call the Company at 1.604.685.9316 at least 48 hours prior to the date of the Meeting for further instructions.

SALAZAR RESOURCES LIMITED

#1305 - 1090 West Georgia Street Vancouver, BC V6E 3V7

NOTICE OF ANNUAL AND SPECIAL MEETING OF SHAREHOLDERS

NOTICE IS HEREBY GIVEN that the Annual and Special Meeting of the Shareholders (the "**Meeting**") of Salazar Resources Limited (the "**Company**") will be held at Suite 1305, 1090 West Georgia Street, Vancouver, British Columbia on Friday, February 17, 2023 at 10:00 a.m. (Pacific time), for the following purposes:

- 1. To receive the audited consolidated financial statements of the Company for the fiscal year ended December 31, 2021, together with the report of the auditors therein;
- 2. To fix the number of directors at six (6);
- 3. To elect directors;
- 4. To appoint D&H Group LLP, Chartered Professional Accountants, as the auditor of the Company for the ensuing year and to authorize the directors to fix the auditor's remuneration;
- 5. To consider and, if thought fit, to pass, with or without variation, an ordinary resolution approving certain amendments to the Company's 10% rolling incentive stock option plan, as more particularly described in the accompanying management information circular; and
- 6. To consider and, if thought fit, to pass, with or without variation, an ordinary resolution of disinterested shareholders to approve certain amendments to the Company's restricted share unit plan and increase the number of restricted share units available for award thereunder to 2,000,000 restricted share units, as more particularly described in the accompanying management information circular.

Accompanying this Notice of Meeting is a Management Information Circular, together with a Form of Proxy and a Request Form for Annual and Interim Financial Statements. The accompanying Management Information Circular provides information relating to the matters to be addressed at the meeting and is incorporated into this Notice of Meeting.

To be valid, the accompanying Form of Proxy, duly completed, dated and signed, must arrive at the office of the Registrar and Transfer Agent of the Company, Computershare Investor Services Inc., not less than 48 hours (excluding Saturdays, Sundays and holidays) before the time for holding the Meeting or delivered to the Chairman of the Meeting on the day of the Meeting prior to the commencement of the Meeting.

If you are a non-registered shareholder of the Company and received this Notice of Meeting and accompanying materials through a broker, a financial institution, a participant, a trustee or administrator of a self-administered retirement savings plan, retirement income fund, education savings plan or other similar self-administered savings or investment plan registered under the Income Tax Act (Canada), or a nominee of any of the foregoing that holds your security on your behalf (the "Intermediary"), please complete and return the materials in accordance with the instructions provided to you by your Intermediary.

COVID-19 GUIDANCE: In light of ongoing concerns related to the spread of COVID-19, and in order to mitigate the potential risks to the health and safety associated with COVID-19, shareholders are strongly encouraged to vote on the matters before the Meeting by proxy rather than attend the meeting in person.

DATED at Vancouver, British Columbia, this 12th day of January, 2023.

By Order of the Board of Directors

"Nick DeMare"

Nick DeMare, Director

SALAZAR RESOURCES LIMITED

#1305 - 1090 West Georgia Street Vancouver, British Columbia V6E 3V7

MANAGEMENT INFORMATION CIRCULAR

(Containing information as at January 12, 2023, unless otherwise stated)

SOLICITATION OF PROXIES

This management information circular ("Information Circular") is furnished in connection with the solicitation of proxies by the management of Salazar Resources Limited (the "Company") for use at the Annual and Special Meeting of Shareholders of the Company (and any adjournment thereof) to be held on February 17, 2023 (the "Meeting") at the time and place and for the purposes set forth in the accompanying Notice of Annual and Special Meeting. While it is expected that the solicitation will be primarily by mail, proxies may be solicited personally or by telephone by the directors, officers and regular employees of the Company at nominal cost. All costs of solicitation by management will be borne by the Company.

The contents and the sending of this Information Circular have been approved by the directors of the Company.

COVID-19

In view of the current COVID-19 outbreak, the Company asks that, in considering whether to attend the Meeting in person, shareholders follow the instructions of the Public Health Agency of Canada at https://www.canada.ca/en/public-health/services/diseases/2019-novel-coronavirus-infection/guidance-documents.html. Shareholders are encouraged not to attend the Meeting in person if experiencing any of the described COVID-19 symptoms of fever, cough or difficulty breathing. The Company may take additional precautionary measures in relation to the Meeting in response to further developments in the COVID-19 outbreak. As always, the Company encourages shareholders to vote prior to the Meeting. Shareholders are encouraged to vote on the matters before the meeting by proxy. Shareholders wishing to attend the Meeting in person must call the Company at 1.604.685.9316 at least 48 hours prior to the date of the Meeting for further instructions.

APPOINTMENT OF PROXYHOLDER

The individuals named in the accompanying form of proxy are directors and/or officers ("Management's Nominees") of the Company. A SHAREHOLDER WISHING TO APPOINT SOME OTHER PERSON (WHO NEED NOT BE A SHAREHOLDER) TO REPRESENT HIM AT THE MEETING HAS THE RIGHT TO DO SO, EITHER BY STRIKING OUT THE NAMES OF THOSE PERSONS NAMED IN THE ACCOMPANYING FORM OF PROXY AND INSERTING THE DESIRED PERSON'S NAME IN THE BLANK SPACE PROVIDED IN THE FORM OF PROXY OR BY COMPLETING ANOTHER FORM OF PROXY. A proxy will not be valid unless the completed form of proxy is received by Computershare Investor Services Inc. (the "Transfer Agent"), Proxy Department, 9th Floor, 100 University Avenue, Toronto, Ontario, M5J 2Y1, not less than 48 hours (excluding Saturdays, Sundays and holidays) before the time for holding the Meeting, or any adjournment thereof. Proxies delivered after that time will not be accepted.

REVOCATION OF PROXIES

A shareholder who has given a proxy may revoke it by an instrument in writing executed by the shareholder or by his attorney duly authorized in writing or, where the shareholder is a company, by a duly authorized officer or attorney of the company, and delivered to the head office of the Company located at Suite 1305, 1090 West Georgia Street, Vancouver, British Columbia V6E 3V7 (Attention: Corporate Secretary), at any time up to and including the last business day preceding the day of the Meeting, or if adjourned, any reconvening thereof, or to the Chairman of the Meeting on the day of the Meeting or, if adjourned, any reconvening thereof or in any other manner provided by law. A revocation of a proxy does not affect any matter on which a vote has been taken prior to the revocation.

INFORMATION FOR NON-REGISTERED SHAREHOLDERS

Only registered shareholders or duly appointed proxyholders are permitted to vote at the Meeting. Most shareholders of the Company are "non-registered" shareholders because the common shares of the Company ("Common Shares") they own are not registered in their names but are instead registered in the names of a brokerage firm, bank or other intermediary or in the name of a clearing agency. Shareholders who do not hold their Common Shares in their own name (referred to herein as "Beneficial Shareholders") should note that only registered shareholders may vote at the Meeting. If Common Shares are listed in an account statement provided to a shareholder by a broker, then in almost all cases those Common Shares will not be registered in such shareholder's name on the records of the Company. Such Common Shares will more likely be registered under the name of the shareholder's broker or an agent of that broker. In Canada, the vast majority of such shares are registered under the name of CDS & Co. (the registration name for CDS Clearing and Depository Services Inc., which company acts as nominee for many Canadian brokerage firms). Common Shares held by brokers (or their agents or nominees) on behalf of a broker's client can only be voted (for or against resolutions) at the direction of the Beneficial Shareholder. Without specific instructions, brokers and their agents and nominees are prohibited from voting shares for the brokers' clients. Therefore, each Beneficial Shareholder should ensure that voting instructions are communicated to the appropriate person well in advance of the Meeting.

Existing regulatory policy requires brokers and other intermediaries to seek voting instructions from Beneficial Shareholders in advance of shareholders' meetings. The various brokers and other intermediaries have their own mailing procedures and provide their own return instructions to clients, which should be carefully followed by Beneficial Shareholders in order to ensure that their Common Shares are voted at the Meeting. Often the form of proxy supplied to a Beneficial Shareholder by its broker is identical to the form of proxy provided by the Company to the registered shareholders. However, its purpose is limited to instructing the registered shareholder (i.e. the broker or agent of the broker) how to vote on behalf of the Beneficial Shareholder. The majority of brokers now delegate responsibility for obtaining instructions from clients to Broadridge Financial Solutions, Inc. ("Broadridge"). Broadridge typically prepares a machine-readable voting instruction form, mails those forms to the Beneficial Shareholders and asks Beneficial Shareholders to return the forms to Broadridge, or otherwise communicate voting instructions to Broadridge (by way of the internet or telephone, for example). Broadridge then tabulates the results of all instructions received and provides appropriate instructions respecting the voting of Common Shares to be represented at the Meeting. A Beneficial Shareholder who receives a Broadridge voting instruction form cannot use that form to vote Common Shares directly at the Meeting. The voting instruction form must be returned to Broadridge (or instructions respecting the voting of Common Shares must be communicated to Broadridge) well in advance of the Meeting in order to have the Common Shares voted.

This Information Circular and accompanying materials are being sent to both registered shareholders and Beneficial Shareholders. Beneficial Shareholders fall into two categories – those who object to their identity being known to the issuers of securities which they own ("OBOs") and those who do not object to their identity being made known to the issuers of the securities they own ("NOBOs"). Subject to the provision of National Instrument 54-101 – Communication with Beneficial Owners of Securities of a Reporting Issuer ("NI 54-101") issuers may request and obtain a list of their NOBOs from intermediaries via their transfer agents. Pursuant to NI 54-101, issuers may obtain and use the NOBO list for distribution of proxy-related materials directly (not via Broadridge) to such NOBOs. If you are a Beneficial Shareholder, and the Company or its agent has sent these materials directly to you, your name, address and information about your holdings of Common Shares have been obtained in accordance with applicable securities regulatory requirements from the intermediary holding the Common Shares on your behalf.

The Company's OBOs can expect to be contacted by Broadridge or their brokers or their broker's agents as set out above.

Although Beneficial Shareholders may not be recognized directly at the Meeting for the purposes of voting Common Shares registered in the name of his broker, a Beneficial Shareholder may attend the Meeting as proxyholder for the registered shareholder and vote the Common Shares in that capacity. Beneficial Shareholders who wish to attend the Meeting and indirectly vote their Common Shares as proxyholder for the registered shareholder should enter their own names in the blank space on the proxy provided to them and return the same to their broker (or the broker's agent) in accordance with the instructions provided by such broker.

All references to shareholders in this Information Circular and the accompanying form of Proxy and Notice of Annual and Special Meeting are to shareholders of record unless specifically stated otherwise.

VOTING OF PROXIES

The shares represented by a properly executed proxy in favour of Management's Nominees as proxyholders in the accompanying form of proxy will:

- (a) be voted or withheld from voting in accordance with the instructions of the person appointing the proxyholder on any ballot that may be taken; and
- (b) where a choice with respect to any matter to be acted upon has been specified in the form of proxy, be voted in accordance with the specification made in such proxy.

ON A POLL, SUCH SHARES WILL BE VOTED **AS DIRECTED BY MANAGEMENT OF THE COMPANY** FOR EACH MATTER FOR WHICH NO CHOICE HAS BEEN SPECIFIED, OR WHERE BOTH CHOICES HAVE BEEN SPECIFIED, BY THE SHAREHOLDER.

The enclosed form of proxy when properly completed and delivered and not revoked confers discretionary authority upon the person appointed proxy thereunder to vote with respect to amendments or variations of matters identified in the Notice of Meeting, and with respect to other matters which may properly come before the Meeting. If any amendments or variations to matters identified in the Notice of Meeting are properly brought before the Meeting or any further or other business is properly brought before the Meeting, it is the intention of the persons designated in the enclosed form of proxy to vote in accordance with their best judgment on such matters or business. At the time of the printing of this Information Circular, the management of the Company knows of no such amendment, variation or other matter that may be presented to the Meeting.

INTEREST OF CERTAIN PERSONS OR COMPANIES IN MATTERS TO BE ACTED UPON

Except as disclosed herein, no person who has been a director or executive officer of the Company at any time since the beginning of the last financial year, nor any proposed nominee for election as a director of the Company, nor any associate or affiliate of any of the foregoing, has any material interest, direct or indirect, by way of beneficial ownership of securities or otherwise, in any matter to be acted upon other than the election of directors, the appointment of auditors or the approval of the proposed restricted share unit plan. Directors and senior officers may, however, be interested in the general authorization granted to the directors with respect to stock options to insiders as detailed herein.

VOTING SECURITIES AND PRINCIPAL HOLDERS OF VOTING SECURITIES

The authorized capital of the Company consists of an unlimited number of Common Shares without par value. As at January 12, 2023 (the "**Record Date**"), the Company had 183,712,079 Common Shares issued and outstanding.

Only shareholders of record at the close of business on the Record Date who either personally attend the Meeting or who have completed and delivered a form of proxy in the manner and subject to the provisions described above shall be entitled to vote or to have their shares voted at the Meeting.

On a show of hands, every individual who is present and is entitled to vote as a shareholder or as a representative of one or more corporate shareholders will have one vote, and on a poll every shareholder present in person or represented by a proxy and every person who is a representative of one or more corporate shareholders, will have one vote for each Common Share registered in that shareholder's name on the list of shareholders as at the Record Date, which is available for inspection during normal business hours at the Transfer Agent and will be available at the Meeting. **Shareholders represented by proxy holders are not entitled to vote on a show of hands.**

To the knowledge of the directors and senior officers of the Company, the following are the only persons or companies who beneficially own, directly or indirectly, or exercise control or direction over, shares carrying more than 10% of the voting rights attached to all outstanding shares of the Company as of the close of business on January 12, 2023:

Name	Number of Shares	<u>Percentage</u>	
Fredy Enrique Salazar*	29.513.529	16.06%	

*Mr. Salazar is the President & Chief Executive Officer of the Company.

ELECTION OF DIRECTORS

The board of directors (the "**Board**") presently consists of six (6) directors and shareholders will be asked at the Meeting to fix the number of directors at six (6) for the ensuing year. It is proposed that six directors be elected for the ensuing year.

The term of office of each of the present directors expires at the Meeting. The persons named below will be presented for election at the Meeting as management's nominees and the persons named by management as proxyholders in the accompanying form of proxy intend to vote for the election of these nominees. Management does not contemplate that any of these nominees will be unable to serve as a director. Each director elected will hold office until the next annual general meeting of the Company or until his successor is elected or appointed, unless his office is earlier vacated in accordance with the Articles of the Company or the provisions of the *Business Corporations Act* (British Columbia) (the "BCBCA").

The following table and notes thereto set out the name of each of management's nominees for election as a director, the province and country in which he/she is ordinarily resident, all offices of the Company now held by him/her, his/her principal occupation, the period of time he/she has been a director of the Company, and the number of shares of the Company beneficially owned by him/her, directly or indirectly, or over which he/she exercises control or direction, as at the date hereof.

Name, Position and Province and Country of Residence ⁽¹⁾	Principal Occupation and, if not at present an elected Director, Occupation during the past five years ⁽¹⁾	Director Since	No. of Shares beneficially held ⁽²⁾
FREDY E. SALAZAR President, Chief Executive Officer & Director Quito, Ecuador	Geologist, principal of Salazar Resources since March 8, 2007. Prior thereto, principal of Congeminpa Ltda, an environmental consulting company from 2001.	Mar. 8/07	29,513,529 (3)
PABLO ACOSTA Chief Financial Officer & Director Quito, Ecuador	Certified Public Accountant in Ecuador. CFO of Salazar Resources since March 8, 2007 and has been a director of several private companies in Ecuador since 1985.	Mar. 8/07	1,094,963
NICK DEMARE ⁽⁴⁾ Director Burnaby, British Columbia	Chartered Professional Accountant; principal of Chase Management Ltd. since 1991 and is a director and/or officer of several publicly listed companies since 1986.	Jun. 6/88	5,729,442 (5)
ETIENNE WALTER ⁽⁴⁾ Director Burnaby, British Columbia	Honorary Consul General of the Republic of Ecuador with jurisdiction over the Provinces of Alberta and B.C., the Northwest Territories and the Yukon Territory.	Mar. 8/07	221,433

Name, Position and Province and Country of Residence ⁽¹⁾	Principal Occupation and, if not at present an elected Director, Occupation during the past five years ⁽¹⁾	Director Since	No. of Shares beneficially held ⁽²⁾
MERLIN MARR-JOHNSON Director & Corporate Secretary United Kingdom	Geologist. Mr. Marr-Johnson has over 25 years' experience in the minerals sector, including work as an exploration geologist for Rio Tinto, an analyst for HSBC and a portfolio manager for Blakeney Management. Mr. Marr-Johnson has worked on projects in South America, Africa, Central Asia and Europe, and as CEO he has brought two companies to AiM, London.	Jan. 24/19	253,000
MARY GILZEAN ⁽⁴⁾ Director Vancouver, British Columbia	Geologist. Ms. Gilzean has over 25 years of experience in international mineral exploration and human resources managementMs. Gilzean has lived and worked in South America, Europe, Africa and North America and has served on the boards of several non-profit organizations in the Vancouver area.	Jun. 14/21	400,000

NOTES:

- (1) The information as to province or state and country of residence and principal occupation, not being within the knowledge of the Company, has been furnished by the respective directors individually.
- (2) The information as to shares beneficially owned or over which a director exercises control or direction, not being within the knowledge of the Company, has been furnished by the respective directors individually.
- (3) Of these shares, 5,045,183 are held by Amlatminas S.A., a private Ecuadorian company controlled by Mr. Salazar.
- (4) Denotes member of Audit Committee.
- Of these shares, 3,219,417 are held through 888 Capital Corp., a private company 50% owned by Mr. DeMare; 888,700 are held through Chase Management Ltd. and 926,325 are held through DNG Capital Corp., private companies wholly owned by Mr. DeMare.

Corporate Cease Trade Orders or Bankruptcies

None of the proposed directors of the Company or any of their personal holding companies:

- (a) is, as at the date of this Information Circular, or has been, within ten years before the date of this Information Circular, a director, chief executive officer or chief financial officer of any Company, including the Company, that:
 - (i) was subject to a cease trade order or similar order or an order that denied the relevant Company access to any exemption under securities legislation, that was in effect for a period of more than 30 consecutive days while that person was acting in the capacity as director, chief executive officer or chief financial officer; or
 - (ii) was subject to a cease trade or similar order or an order that denied the company access to any exemption under securities legislation, that was in effect for a period of more than 30 consecutive days, that was issued after the person ceased to be a director, chief executive officer or chief financial officer of the Company and which resulted from an event that occurred while that person was acting in the capacity as director, chief executive officer or chief financial officer; or
- (b) is as at the date of this Information Circular or has been within the 10 years before the date of this Information Circular, a director or executive officer of any Company, including the Company, that

while that person was acting in that capacity, or within a year of that person ceasing to act in that capacity, became bankrupt, made a proposal under any legislation relating to bankruptcy or insolvency or was subject to or instituted any proceedings, arrangement or compromise with creditors or had a receiver, receiver manager or trustee appointed to hold its assets; or

(c) has, within the 10 years before the date of this Information Circular, become bankrupt, made a proposal under any legislation relating to bankruptcy or insolvency, or become subject to or instituted any proceedings, arrangements or compromise with creditors, or had a receiver, receiver manager or trustee appointed to hold the assets of that person.

None of the proposed directors or any of their personal holding companies has been subject to:

- (a) any penalties or sanctions imposed by a court relating to securities legislation or by a securities regulatory authority or has entered into a settlement agreement with a securities regulatory authority; or
- (b) any other penalties or sanctions imposed by a court or regulatory body that would likely be considered important to a reasonable securityholder in deciding whether to vote for a proposed director.

STATEMENT OF EXECUTIVE COMPENSATION

The following disclosure regarding executive compensation is provided pursuant to National Instrument 51-102 Continuous Disclosure and in accordance with Form 51-102F6V for venture issuers.

General

For the purposes of this Information Circular:

- "compensation securities" includes stock options, convertible securities, exchangeable securities and similar
 instruments including stock appreciation rights, deferred share units and restricted stock units granted or issued
 by the company or one of its subsidiaries for services provided or to be provided, directly or indirectly, to the
 company or any of its subsidiaries;
- "external management company" includes a subsidiary, affiliate or associate of the external management company;
- "named executive officer" or "NEO" means each of the following individuals:
 - (a) each individual who, in respect of the company, during any part of the most recently completed financial year, served as chief executive officer, including an individual performing functions similar to a chief executive officer;
 - (b) each individual who, in respect of the company, during any part of the most recently completed financial year, served as chief financial officer, including an individual performing functions similar to a chief financial officer:
 - (c) in respect of the company and its subsidiaries, the most highly compensated executive officer other than the individuals identified in paragraphs (a) and (b) at the end of the most recently completed financial year whose total compensation was more than \$150,000, as determined in accordance with subsection 1.3(5), for that financial year;
 - (d) each individual who would be a named executive officer under paragraph (c) but for the fact that the individual was not an executive officer of the company, and was not acting in a similar capacity, at the end of that financial year;
- "plan" includes any plan, contract, authorization, or arrangement, whether or not set out in any formal document, where cash, compensation securities or any other property may be received, whether for one or more persons;

• "underlying securities" means any securities issuable on conversion, exchange or exercise of compensation securities.

During the financial years ended December 31, 2021 and 2020, the Company had two NEOs: Fredy Salazar, President and CEO, and Pablo Acosta, CFO.

Director and NEO Compensation, Excluding Options and Compensation Securities

The following table of compensation, excluding options and compensation securities, provides a summary of the compensation paid by the Company to each NEO and director of the Company, current or former, for the completed financial years ended December 31, 2021 and 2020. Options and compensation securities are disclosed under the heading "Stock Options and Other Compensation Securities and Instruments" below.

Table of Compensation, Excluding Compensation Securities							
Name and position	Year ⁽¹⁾	Salary, consulting fee, retainer or commission (\$)^{(2)}	Bonus (\$) ⁽²⁾	Committee or meeting fees (\$) ⁽²⁾	Value of perquisites (\$)(2)	Value of all other compensation (\$)(2)	Total compensation (\$) ⁽²⁾
Fredy Salazar President, CEO and Director	2021 2020	132,754 102,119	-		-	448,140 ⁽³⁾ 304,635 ⁽³⁾	580,894 406,754
Pablo Acosta CFO and Director	2021 2020	127,053 115,879	-		-	171,972 ⁽⁴⁾ 167,377 ⁽⁴⁾	299,025 283,256
Merlin Marr-Johnson ⁽⁵⁾ Executive Vice- President, Corporate Secretary and Director	2021 2020	132,000 125,750	-		-		132,000 125,750
Nick DeMare Director and Assistant Corporate Secretary	2021 2020	45,134 48,260	- -		-	62,039 ⁽⁶⁾ 56,055 ⁽⁶⁾	107,173 104,315
Etienne Walter Director	2021 2020	22,508 24,111	-		-	-	22,508 24,111
Mary Gilzean ⁽⁷⁾ Director	2021 2020	12,423 N/A	- N/A	- N/A	- N/A	- N/A	12,423 N/A
Jennifer Wu ⁽⁸⁾ Former Director	2021 2020	7,064 24,111	-		-	-	7,064 24,111

NOTES:

- (1) Financial years ended December 31.
- (2) All amounts shown were paid in Canadian currency, the reporting currency of the Company.
- (3) Includes \$202,786 (2020 \$148,202) paid for professional services and \$240,710 (2020 \$151,465) for equipment rental services provided by Amlatminas S.A., a private corporation owned by Mr. Salazar, and \$4,644 (2020 \$4,968) health benefits premiums paid.
- (4) Includes \$171,972 (2020 \$167,377) paid for professional services provided by corporations owned by Mr. Acosta and \$1,868 (2020 \$1,999) health benefits premiums paid.
- (5) Mr. Marr-Johnson was appointed as a director on January 14, 2019 and as Executive Vice-President on October 14, 2021.
- (6) Paid to Chase Management Ltd. ("Chase"), a private corporation owned by Mr. DeMare, for bookkeeping, accounting and corporate services rendered by Chase personnel exclusive of Mr. DeMare.
- (7) Ms. Gilzean was appointed as a director on June 14, 2021.
- (8) Ms. Wu resigned as a director on April 23, 2021.

Stock Options and Other Compensation Securities and Instruments

The following table of compensation securities provides a summary of compensation securities granted or issued by the Company to the NEOs and directors of the Company, current or former, for the financial year ended December 31, 2021, for services provided or to be provided, directly or indirectly, to the Company or any of its subsidiaries:

	Compensation Securities							
Number of compensation securities, number of underlying securities, and percentage Name and Position Number of compensation security Type of compensation percentage of class Type of compensation percentage issue or price date of grant year end (\$) (\$) (\$) (\$) (\$) (\$) (\$)								
Fredy Salazar	Options ⁽²⁾	520,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	180,000	Apr 6/21	N/A	0.29	0.295	N/A	
Pablo Acosta	Options ⁽⁴⁾	160,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	80,000	Apr 6/21	N/A	0.29	0.295	N/A	
Merlin Marr-Johnson	Options ⁽⁵⁾	400,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	150,000	Apr 6/21	N/A	0.29	0.295	N/A	
Nick DeMare	Options ⁽⁶⁾	104,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	80,000	Apr 6/21	N/A	0.29	0.295	N/A	
Etienne Walter	Options ⁽⁷⁾	100,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	35,000	Apr 6/21	N/A	0.29	0.295	N/A	
Mary Gilzean	Options ⁽⁸⁾	500,000	Jun 14/21	\$0.37	0.37	0.295	Jun 14/26	
	RSUs ⁽³⁾	100,000	Jun 14/21	N/A	0.37	0.295	N/A	
Jennifer Wu	Options ⁽⁹⁾	100,000	Apr 6/21	\$0.29	0.29	0.295	Apr 6/26	
	RSUs ⁽³⁾	35,000	Apr 6/21	N/A	0.29	0.295	N/A	

NOTES:

- (1) Market value of the securities underlying the options on December 30, 2021, being the last trading day of the Company's shares for the financial year.
- (2) As at December 31, 2021, Mr. Salazar held 1,470,000 stock options of the Company entitling him to acquire, upon exercise, 1,470,000 common shares in the capital of the Company. As at December 31, 2021, 950,000 these stock options were vested.
- (3) The RSUs granted on April 6, 2021 and June 14, 2021 will vest 100% two years from the date of grant.
- (4) As at December 31, 2021, Mr. Acosta held 510,000 stock options of the Company entitling him to acquire, upon exercise, 510,000 common shares in the capital of the Company. As at December 31, 2021, 350,000 of these stock options were vested.
- (5) As at December 31, 2021, Mr. Marr-Johnson held 1,400,000 stock options of the Company entitling him to acquire, upon exercise, 1,400,000 common shares in the capital of the Company. As at December 31, 2021, 1,000,000 of these stock options were vested.
- (6) As at December 31, 2021, Mr. DeMare held 654,000 stock options of the Company entitling him to acquire, upon exercise, 654,000 common shares in the capital of the Company. These include 150,000 stock options granted to Chase. As at December 31, 2021, 550,000 of these stock options were vested.
- (7) As at December 31, 2021, Mr. Walter held 400,000 stock options of the Company entitling him to acquire, upon exercise, 400,000 common shares in the capital of the Company. As at December 31, 2021, 300,000 of these stock options were vested.
- (8) As at December 31, 2021, Ms. Gilzean held 500,000 stock options of the Company entitling her to acquire, upon exercise, 500,000 common shares in the capital of the Company. As at December 31, 2021 none of these stock options were vested.
- (9) As at December 31, 2021, Ms. Wu held nil stock options. Ms. Wu resigned as a director on April 23, 2021 and exercised all 900,000 of her vested options on August 19, 2021. On September 30, 2021, 100,000 of her unvested options expired without exercise.

The following table provides a summary of each exercise of compensation securities by each NEO and director of the Company for the financial year ended December 31, 2021:

	Exercise of Compensation Securities							
Number of Type of underlying compensation security Name Name Number of underlying exercise price per security on date of security exercise (\$) Name Difference between exercise price per security on date of price on date on exercise (\$) (\$) (\$) Difference between exercise price on date of price on date on exercise (\$)								
Fredy Salazar	Options Options	148,514 1,151,486	0.14 0.14	Apr 7/21 Nov 19/21	0.305 0.38	0.165 0.24	24,505 276,357	
Pablo Acosta	Options	900,000	0.14	Nov 30/21	0.39	0.25	225,000	
Merlin Marr-Johnson	Options	Nil	N/A	N/A	N/A	N/A	N/A	
Nick DeMare	Options Options	200,000 ⁽¹⁾ 700,000	0.14 0.14	Nov 16/21 Nov 29/21	0.33 0.39	0.19 0.25	38,000 175,000	
Etienne Walter	Options	600,000	0.14	Dec 1/21	0.305	0.165	99,000	
Mary Gilzean	Options	Nil	N/A	N/A	N/A	N/A	N/A	
Jennifer Wu	Options Options	600,000 300,000	0.14 0.135	Aug 19/21 Aug 19/21	0.34 0.34	0.20 0.205	120,000 61,500	

NOTE:

(1) Exercised by Chase.

Stock Option Plans and Other Incentive Plans

A. 10% Rolling Stock Option Plan

The Company has in place a 10% rolling stock option plan (the "Option Plan"). The purpose of the Option Plan is to provide the Company with a share related mechanism to enable the Company to attract, retain and motivate qualified directors, officers, employees and other service providers, to reward directors, officers, employees and other service providers for their contribution toward the long term goals of the Company and to enable and encourage such individuals to acquire shares of the Company as long term investments.

The following is a brief description of the Option Plan:

- (a) To be eligible for the issuance of a stock option under the Option Plan an optionee must either be a director, employee (including an officer) or consultant of the Company or any subsidiary of the Company or a company owned by an employee, director or consultant at the time the option is granted. Options may be granted only to an individual or to a company that is owned by individuals eligible for an option grant.
- (b) The options granted pursuant to the Option Plan will be exercisable at a price which is not lower than the market value of the Company's Shares at the time the option is granted less any applicable discounts permitted by the applicable regulatory authorities. "Market Value" will be the closing trading price of the Company's Shares on the TSX Venture Exchange (the "Exchange") or such other stock exchange upon which the common shares are listed on the trading day immediately preceding the date of the grant of the option.
- (c) Options granted under the Option Plan will be granted for a term not to exceed ten years from the date of their grant. All options will terminate on the earlier of the expiry of their term and the date of termination of an option holder's employment, engagement or position with the Company if terminated for just cause or on other bases as

set out in the Option Plan, otherwise 90 days following termination of employment or cessation of the option holder's position with the Company.

- (d) The Company's Board may, at their discretion, impose vesting provisions on Options granted under the Option Plan. Notwithstanding any vesting schedule to which Options are subject, Options shall cease to vest immediately if the employment or engagement of an Option Holder as an Employee or Consultant or the position of an Option Holder as a director or officer of the Company or a Subsidiary is terminated for any reason whatsoever. In which case, the Option Holder may only exercise such number of Options that are vested as at the date of termination of such Option Holder's employment, engagement or appointment as a director or officer.
- (e) Options will also be non-assignable and non-transferable; provided that they will be exercisable by an option holder's legal heirs or personal representatives, subject to the expiry date of such option, for up to 12 months following the death or termination of an option holder due to disability, and up to 12 months following the death of an option holder terminated for disability within the previous 12 months.
- (f) The number of common shares of the Company reserved for issuance to any one person on a yearly basis cannot exceed 5% of the number of issued and outstanding common shares of the Company at the time of the grant of options, unless the Company has obtained disinterested shareholder approval as required by the Exchange. The aggregate number of options granted to employees or consultants engaged in investor relations activities must not exceed 2% of the outstanding issue in any 12 month period and such options must vest in stages over 12 months with no more than 25% of the options vesting in any three month period.
- (g) In the event that the expiration date of Options granted under the Option Plan falls in a period during which the Company has imposed a restriction on its directors, officers, employees and consultants from trading in securities of the Company, the expiry date of such Options will be extended for a period of time ending on the tenth business day after the expiry of the Black-Out to provide such Option Holders with an extension to the right to exercise such Options, so long as the expiry date does not exceed ten years from the date of grant of such Options.
- (h) If a material alteration in the capital structure of the Company occurs as a result of a consolidation, subdivision, conversion, exchange, reclassification or otherwise, the Board shall make adjustments to the Option Plan and to the options then outstanding under it as the Board determines to be appropriate and equitable under the circumstances, unless the Board determines that it is not practicable or feasible to do so, in which event the options granted under the Option Plan will terminate as set forth above.
- (i) The Board may amend the terms of the Option Plan or the terms and conditions of any option thereafter to be granted, subject to approval of any stock exchange on which the Company is listed, provided that where such amendment relates to an existing option and it would materially decrease the rights or benefits accruing to an option holder or materially increase the obligations of an option holder, then, unless otherwise excepted out by a provision of the Option Plan, the Board must also obtain the written consent of the option holder in question to such amendment. If at the time the exercise price of an option is reduced the option holder is an insider of the Company, the insider must not exercise the option at the reduced exercise price until the reduction in exercise price has been approved by the disinterested shareholders of the Company.

As at the date of this information circular, the Company had 183,712,079 Shares issued and outstanding so that a maximum of 18,371,207 Shares would be available for issuance pursuant to the stock options granted under the Option Plan. There are currently 6,887,000 stock options outstanding leaving 11,484,207 Shares available for grant of options under the Option Plan.

The Company, subject to Exchange acceptance and shareholder approval, amended its Option Plan. Shareholders will also be asked at the Meeting to pass an ordinary resolution approving the Option Plan Amendment Resolution. See "Particulars of Matters to be Acted Upon – Approval of Option Plan Amendments".

B. Fixed Restricted Share Unit Plan

The Company has in place a fixed restricted share unit plan (the "RSU Plan"). The RSU Plan described below was adopted by the Company on August 27, 2020 and approved by the shareholders at the annual general meeting held on September 25, 2020.

Restricted share units ("RSUs") granted under the RSU Plan will rise and fall in value based on the value of the Common Shares. Unlike Options, RSUs will not require the payment of any monetary consideration to the Company. Instead, each RSU represents a right to receive one Common Share following the attainment of vesting criteria determined at the time of the award.

The RSU Plan is a fixed plan pursuant to which the number of Common Shares that may be issued pursuant to RSUs granted under the RSU Plan is fixed at 1,000,000; provided, however, that the total number of Common Shares which may be issued pursuant to RSUs and Options granted under the Option Plan is a maximum of 10% of the issued and outstanding Common Shares at the time of grant. The Company awarded a total of 898,000 RSUs during the Company's last completed financial year ended December 31, 2021.

The following information is intended to be a brief description of the material terms of the RSU Plan and is qualified in its entirety by the full text of the RSU Plan, a copy of which is attached as Schedule "B" to the Information Circular filed on SEDAR on August 25, 2020.

- (a) Eligible Persons: The Board of Directors or a committee delegated by the Board of Directors under the RSU Plan (the "Committee") may grant RSUs to directors, officers, employees or consultants of the Company or a subsidiary of the Company (the "Participants") provided that the Board, together with such individuals or companies, are responsible for ensuring and confirming that such person is a bona fide Participant.
- (b) Fixed Plan: The RSU Plan is a fixed plan, such that the aggregate number of Common Shares that may be issued pursuant to the Plan shall not exceed 1,000,000 Common Shares, subject to the number of Common Shares reserved for issuance under the Option Plan and the RSU Plan being no greater than 10% of the Company's issued and outstanding Common Shares at any time.
- (c) Vesting: Each RSU will vest in such manner as determined by the Board of Directors or the Committee at the time of grant.
- (d) Settlement of RSUs: On the vesting date, the Company at its sole and absolute discretion have the option of settling the RSUs in cash (if applicable), Common Shares acquired by the Company on the TSX Venture Exchange (the "Exchange") or Common Shares to be issued from the treasury of the Company.
- (e) Limitations: The RSU Plan includes the following additional limitations: (i) the number of Common Shares reserved for issuance to Participants retained to provide Investor Relations Activities (as defined under the policies of the Exchange) under all security based compensation arrangements in any 12 month period shall not exceed 2% of the issued and outstanding Common Shares; (ii) the number of Common Shares reserved for issuance to any one Participant retained as a consultant to provide services to the Company or its subsidiaries under all security based compensation arrangements in any 12 month period shall not exceed 2% of the issued and outstanding Common Shares; (iii) the number of Common Shares reserved for issuance to any one Participant under all security based compensation arrangements in any 12 month period will not exceed 5% of the issued and outstanding Common Shares; (iv) unless the Company has received disinterested shareholder approval to do so, the number of Common Shares issuable to insiders, at any time, under all security based compensation arrangements, shall not exceed 10% of the issued and outstanding Common Shares; and (v) unless the Company has received disinterested shareholder approval to do so the number of Common Shares issued to insiders, within any one year period, under all security based compensation arrangements, shall not exceed 10% of the issued and outstanding Common Shares.
- (f) Ceasing to be a director, officer, employee or consultant: The RSU Plan provides that that if a Participant shall cease to be a director or officer of or be in the employ of, or a consultant or other Participant to, the Company or a subsidiary for any reason whatsoever including, without limitation, retirement, resignation or involuntary termination (with or without cause), as determined by the Board of Directors in its sole discretion, before all of the awards respecting RSUs credited to the Participant's account have vested or are forfeited pursuant to any other provision hereof, (i) such Participant shall cease to be a Participant as of the forfeiture date, (ii) the former Participant shall forfeit all unvested awards respecting RSUs credited to the Participant's account effective as at the forfeiture date, (iii) any award value corresponding to any vested RSUs remaining unpaid as of the forfeiture

date shall be paid to the former Participant and (iv) the former Participant shall not be entitled to any further payment from the RSU Plan.

- (g) Change of control: In the event of a Change of Control (as defined in the RSU Plan), the Board or the Committee shall have absolute discretion to determine if all issued and outstanding RSUs shall vest (whether or not then vested) upon the Change of Control and the vesting date shall be the date which is immediately prior to the time such Change of Control takes place, or at such earlier time as may be established by the Board of Directors or the Committee, in its absolute discretion, prior to the time such Change of Control takes place.
- (h) Transferability: Except as required by law, the rights of a Participant under the RSU Plan are not capable of being assigned, transferred, alienated, sold, encumbered, pledged, mortgaged or charged and are not capable of being subject to attachment or legal process for the payment of any debts or obligations of the Participant.
- (i) Amendments: The Board of Directors may amend the RSU Plan in any way, or discontinue the RSU Plan altogether, and may amend, in any way, any RSU granted under the RSU Plan at any time without the consent of a Participant, provided that such amendment shall not adversely alter or impair any RSU previously granted under the RSU Plan or any related RSU agreement, except as otherwise permitted under the RSU Plan. In addition, the Board of Directors may, by resolution, make any amendment to the RSU Plan or any RSU granted under it (together with any related RSU agreement) without shareholder approval, provided however, that the Board will not be entitled to amend the RSU Plan or any RSU granted under it without shareholder (disinterested shareholder approval if applicable) and, if applicable, Exchange approval, in order to: (i) increase the maximum number of Common Shares issuable pursuant to the RSU Plan; (ii) cancel an RSU and subsequently issue to the holder of such RSU a new RSU in replacement thereof; (iii) extend the term of an RSU, but not beyond the Expiry Date; (iv) permit the assignment or transfer of an RSU other than as provided for in the RSU Plan; (v) add to the categories of persons eligible to participate in this Plan; or (viii) in any other circumstances where Exchange and shareholder approval is required by the Exchange. Any renewal of the RSU Plan will be subject to disinterested shareholder approval, and Exchange approval as applicable.

On December 6, 2022, the Board approved certain amendments to the RSU Plan, which include the increase in the maximum number of Common Shares which may be issued under the RSU Plan from 1,000,000 Common Shares to 2,000,000 Common Shares (including Common Shares already issued under the RSU Plan), subject to shareholder approval and stock exchange approvals. See "Particulars of Other Matters to be Acted Upon - Amendment of Restricted Share Unit Plan".

Employment, Consulting and Management Agreements

Management services are provided to the Company by Chase Management Ltd. ("Chase") of Suite 1305, 1090 West Georgia Street, Vancouver, BC V6E 3V7 pursuant to an arrangement with the Company on a month-to-month basis. Chase is a corporation controlled by Nick DeMare, a director of the Company. During the year ended December 31, 2021, a total of \$107,173 was billed by Chase, as to \$62,039 in accounting, management and administration services provided by Chase personnel and \$45,134 for Mr. DeMare's services. The amounts have been recorded at the exchange amounts agreed to by Chase and the Company.

Oversight and Description of Director and NEO Compensation

Compensation, Philosophy and Objectives

The Board meets to discuss and determine management compensation, without reference to formal objectives, criteria or analysis. The general objectives of the Company's compensation strategy are to (a) compensate management in a manner that encourages and rewards a high level of performance and outstanding results with a view to increasing long-term shareholder value; (b) align management's interests with the long-term interests of shareholders; and (c) ensure that the total compensation package is designed in a manner that takes into account the constraints that the Company is under by virtue of the fact that it is a junior mineral exploration company without a history of earnings.

The Board, as a whole, ensures that total compensation paid to all Named Executive Officers (or NEOs), as hereinafter defined, is fair and reasonable. The Board, as a whole, recommends levels of executive compensation that are competitive, motivating and commensurate with the time spent by executive officers in meeting their obligations. While the Board

does not have direct experience related to executive compensation, the Board relies on their experience as officers and directors with other junior mining companies in assessing compensation levels.

Analysis of Elements

Base salary is used to provide the Named Executive Officers a set amount of money during the year with the expectation that each Named Executive Officer will perform his responsibilities to the best of his ability and in the best interests of the Company.

The Company considers the granting of incentive stock options to be a significant component of executive compensation as it allows the Company to reward each NEO's efforts to increase value for shareholders without requiring the Company to use cash from its treasury. Stock options are generally awarded to executive officers at the commencement of employment and periodically thereafter. The terms and conditions of the Company's stock option grants, including vesting provisions and exercise prices, are governed by the terms of the Company's Option Plan. A description of the significant terms of the Option Plan is found under the heading "Stock Option Plans and Other Incentive Plans".

The Company does not determine executive compensation based on the share price performance. Overall the salaries or consulting fees payable to the NEOs, in particular to the Company's CEO, have had a minor upward trend in order to provide competitive levels of compensation necessary to attract and maintain executive talent.

The Board has considered the implications of the risks associated with the Company's compensation practices. The Board acknowledges that the Company, as a junior natural resource company, does not presently generate any revenues, and that all management compensation to date has been derived solely from cash in the Company's treasury, acquired by way of equity financings to date, and the grant of incentive stock options to directors, management, contractors and employees. Salary compensation to the Named Executive Officers is provided for under verbal understandings or written consulting agreements with the Named Executive Officers or management companies under their control. Upon the occurrence of certain events, the Company's early termination of these contracts may also trigger additional balloon payments, which could adversely impact the Company's working capital.

Option-based Awards

As described above, the Company has in place an Option Plan. The Company's directors, employees, officers and certain consultants are entitled to participate in the Option Plan. The Option Plan is designed to encourage share ownership and entrepreneurship on the part of the senior management and other employees. The Board believes that the Option Plan aligns the interests of the NEOs and the Board with shareholders by linking a component of executive compensation to the longer term performance of the Company's common shares.

Options are granted by the Board. In monitoring or adjusting the option allotments, the Board takes into account its own observations on individual performance (where possible) and its assessment of individual contribution to shareholder value, previous option grants and the objectives set for the NEOs and the Board. The scale of options is generally commensurate to the appropriate level of base compensation for each level of responsibility.

In addition to determining the number of options to be granted pursuant to the methodology outlined above, the Board also makes the following determinations:

- parties who are entitled to participate in the Option Plan;
- the exercise price for each stock option granted, subject to the provision that the exercise price cannot be lower than the prescribed discount permitted by the TSX Venture Exchange (the "Exchange") from the market price on the date of grant;
- the date on which each option is granted;
- the vesting period, if any, for each stock option;

- the other material terms and conditions of each stock option grant; and
- any re-pricing or amendment to a stock option grant.

The Board makes these determinations subject to and in accordance with the provisions of the Option Plan. The Board reviews and approves grants of options on an annual basis and periodically during a financial year.

There is no restriction on NEOs or Directors regarding the purchase of financial instruments, including prepaid variable forward contracts, equity swaps, collars or units or exchange funds that are designed to hedge or offset a decrease in market value of equity securities granted as compensation or held directly or indirectly by the NEO or Director for the financial year ended December 31, 2021.

No NEO or Director, directly or indirectly, purchased any financial instruments or employed a strategy to hedge or offset a decrease in market value of equity securities granted as compensation or held.

Pension

The Company does not have any form of pension plan that provides for payments or benefits to the NEOs at, following, or in connection with retirement. The Company does not have any form of deferred compensation plan.

SECURITIES AUTHORIZED FOR ISSUANCE UNDER EQUITY COMPENSATION PLANS

The following table provides information regarding compensation plans under which securities of the Company are authorized for issuance to directors, officers, employees and consultants in effect as of the end of the Company's most recently completed fiscal year end:

Plan Category	Column (a) Number of Securities to be Issued Upon Exercise of Outstanding Options, Warrants and Rights	Column (b) Weighted-Average Exercise Price of Outstanding Options, Warrants and Rights	Column (c) Number of Securities Remaining Available for Future Issuance Under Equity Compensation Plans (Excluding Securities Reflected in Column (a))
Equity Compensation Plans	6,937,000 Options	0.19	8,329,207 Options ⁽¹⁾
Approved by Securityholders	863,000 RSUs	N/A	137,000 RSUs ⁽²⁾
Equity Compensation Plans Not Approved by Securityholders	-	-	-
Total	6,937,000 Options	0.19	8,329,207 Options ⁽¹⁾
	863,000 RSUs	N/A	137,000 RSUs ⁽²⁾

NOTES:

- (1) Based upon the Company having 152,662,073 common shares issued and outstanding as at December 31, 2021. The Company currently has in place a "rolling" stock option plan whereby the maximum number of common shares that may be reserved for issuance pursuant to such plan will not exceed 10% of the issued shares of the Company at the time of the stock option grant. See "Stock Option Plans and Other Incentive Plans 10% Rolling Stock Option Plan" for further particulars of the stock option plan.
- (2) In September 2020 the Company adopted a restricted share unit plan whereby the maximum number of restricted share units ("RSUs") that may be reserved for issuance pursuant to such plan will not exceed 1,000,000 common shares, subject to the number of common shares reserved for issuance pursuant to such plan and the stock option plan not exceeding 10% of the issued shares of the Company at any time. See "Stock Option Plans and Other Incentive Plans Restricted Share Unit Plan" for further particulars of the restricted share unit plan.

The Company has amended its Option Plan, subject to receipt of Exchange acceptance and shareholder approval. Shareholders will be asked at the Meeting to pass an ordinary resolution approving the Option Plan Amendment Resolution. See "Particulars of Matters to be Acted Upon –Approval of Option Plan Amendments".

The Company has also amended its RSU Plan, subject to receipt of Exchange acceptance and disinterested shareholder approval. Shareholders will be asked at the Meeting to pass an ordinary resolution approving the RSU Plan Amendment Resolution. See "Particulars of Matters to be Acted Upon – Amendment of Restricted Share Unit Plan".

INDEBTEDNESS OF DIRECTORS AND EXECUTIVE OFFICERS

As of January 12, 2023, the date of this Information Circular, no director, executive officer, employee, proposed management nominee for election as a director of the Company, nor any associate of any such director, executive officer, or proposed management nominee of the Company, or any former director, executive officer or employee of the Company or any of its subsidiaries, was indebted to the Company or any of its subsidiaries, or indebted to another entity where such indebtedness was the subject of a guarantee, support agreement, letter of credit or other similar arrangement or understanding provided by the Company or any of its subsidiaries.

INTEREST OF INFORMED PERSONS IN MATERIAL TRANSACTIONS

Except as disclosed below, none of the proposed directors, directors or executive officers of the Company, a director or executive officer of a person or company that is itself an informed person (as defined in National Instrument 51-102 *Continuous Disclosure Obligations*) or subsidiary of the Company, nor any person or company who beneficially owns, or controls or directs, directly or indirectly, Common Shares or a combination of both, carrying more than 10% of the voting rights attached to the outstanding Common Shares nor an associate or affiliate of any of the foregoing persons has since January 1, 2021 (being the commencement of the Company's last completed financial year) any material interest, direct or indirect, in any transactions which materially affected or would materially affect the Company or any of its subsidiaries.

During fiscal 2021 the Company incurred \$240,710 for equipment rental services and \$202,786 for professional services provided by Amlatminas S.A., a private corporation controlled by Mr. Salazar and Mr. Acosta.

During fiscal 2021 the Company incurred \$34,620 for storage rental provided by Agrosamex S.A., a private corporation controlled by the son of Mr. Salazar.

During fiscal 2021 the Company incurred \$167,459 for environmental services provided by Cinge CIA LTDA, a private corporation controlled by Mr. Acosta.

During fiscal 2021 the Company incurred \$4,513 for geological services provided by Sthejobs Services S.A., a private corporation controlled by Mr. Acosta.

APPOINTMENT OF AUDITOR

Unless such authority is withheld, the persons named in the accompanying proxy intend to vote for the appointment of D&H Group LLP, Chartered Professional Accountants, as the auditor of the Company and to authorize the directors to fix their remuneration.

AUDIT COMMITTEE

Under National Instrument 52-110 – *Audit Committees* ("NI 52-110"), companies are required to provide disclosure with respect to their audit committee, including the text of the audit committee's charter, the composition of the audit committee and the fees paid to the external auditor.

Audit Committee Charter

The Company has adopted a Charter of the Audit Committee of the Board of Directors, the text of which is set out in the attached Schedule "A" to this Information Circular.

Composition of the Audit Committee

Following the election of the directors pursuant to this Information Circular, it is intended that the following will be the members of the Audit Committee:

Nick DeMare	Independent ⁽¹⁾	Financially literate ⁽²⁾
Etienne Walter	Independent ⁽¹⁾	Financially literate ⁽²⁾
Mary Gilzean	Independent ⁽¹⁾	Financially literate ⁽²⁾

NOTES:

- (1) A member of an audit committee is independent if the member has no direct or indirect material relationship with the Company that could, in the view of the Board of Directors, reasonably interfere with the exercise of a member's independent judgment.
- (2) An individual is financially literate if he has the ability to read and understand a set of financial statements that present a breadth of complexity of accounting issues that are generally comparable to the breadth and complexity of the issues that can reasonably be expected to be raised by the Company's financial statements.

Relevant Education and Experience

Mr. Nick DeMare is a Chartered Professional Accountant and has been the President of Chase Management Ltd. since 1991. Chase is a private company which provides accounting, management, securities regulatory compliance and corporate secretarial services to companies listed on the Toronto and Venture Exchanges and their predecessors. He also serves as an officer and/or director of a number of public companies listed on the Toronto and Venture Exchanges. He holds a Bachelor of Commerce degree from the University of British Columbia and is a member in good standing with the Institute of Chartered Professional Accountants of British Columbia.

Mr. Etienne Walter has been an Honorary Consul General of Republic of Ecuador with jurisdiction over the Provinces of Alberta and British Columbia, the Northwest Territories and Yukon Territory since December 1993 and has been recognized by the Canadian Government since March 17, 1994. Mr. Walter has worked with many public and private corporations, consulting on matters of international relations and trade, particularly in the resource and energy fields, throughout South America. He has also served as a director of other public and private companies.

Ms. Mary Gilzean has over 25 years of experience in international mineral exploration and human resources management. She has lived and worked in Africa, Europe, and North and South America with Teck Resources and BHP Billiton. She is a graduate in geology from Stanford University and holds a Master's Degree in geology from the University of California, Berkeley. She has served on the boards of several non-profit organizations in the Vancouver area.

As such, each proposed member of the audit committee has acquired knowledge and understanding of the financial issues and accounting principles that are relevant in assessing this Company's financial disclosures and internal control systems.

Audit Committee Oversight

At no time since the commencement of the Company's most recently completed financial year was a recommendation of the Committee to nominate or compensate an external auditor not adopted by the Board of Directors.

Reliance on Certain Exemptions

At no time since the commencement of the Company's most recently completed financial year has the Company relied on the exemption in Section 2.4 of NI 52-110 (De Minimis Non-audit Services), the exemptions in Subsection 6.1.1(4) (Circumstances Affecting the Business or Operations of the Venture Issuer), Subsection 6.1.1(5) (Events Outside Control of Member), Subsection 6.1.1(6) (Death, Incapacity or Resignation) or an exemption from NI 52-110, in whole or in part, granted under Part 8 of NI 52-110.

Pre-Approval Policies and Procedures

The Committee is required to review the performance of the Company's external auditors and to approve in advance provision of services other than auditing. The Committee is also required to consider the independence of the external auditors, including reviewing the range of services provided in the context of all consulting services bought by the

Company. The Committee is authorized to approve any non-audit services or additional work that the Chairman deems necessary. In such a case, the Chairman of the Committee is to notify the other members of the Committee of such non-audit or additional work.

External Auditor Service Fees (By Category)

The aggregate fees billed by the Company's external auditors in each of the last two fiscal years for audit fees are as follows:

Financial Year Ending	Audit Fees ⁽¹⁾	Audit Related Fees(2)	Tax Fees ⁽³⁾	All Other Fees ⁽⁴⁾
2021	\$154,486	Nil	Nil	Nil
2020	\$90,317	Nil	\$6,000	\$1,358

NOTES:

- (1) The aggregate audit fees billed.
- (2) The aggregate fees billed for assurance and related services that are reasonably related to the performance of the audit or the review of the Company's financial statements are not included under the heading of "Audit Fees".
- (3) The aggregate fees billed for professional services rendered for tax compliance, tax advice and tax planning, as well as the preparation of corporate income tax returns.
- (4) The aggregate fees billed for products and services other than as set out under the headings "Audit Fees", "Audit Related Fees" and "Tax Fees".

Exemption

The Company is currently a "Venture Issuer", as defined in Section 1.1 of NI 52-110. Accordingly, in providing the disclosure contained in this section titled "Audit Committee", the Company is relying upon the exemption in Section 6.1 of NI 52-110 (which is available to all Venture Issuers).

CORPORATE GOVERNANCE PRACTICE

National Instrument 58-101, *Disclosure of Corporate Governance Practices* ("NI 58-101") requires issuers to disclose their governance practices in accordance with that instrument. The Company is a "venture issuer" within the meaning of NI 58-101.

The Board has adopted certain corporate governance policies to reflect the Company's commitment to good corporate governance, and to comply with NI 58-101, Form 58-101F2 - Corporate Governance Disclosure (Venture Issuers) and National Policy 58-201 - Corporate Governance Guidelines. The Board periodically reviews these policies and proposes modifications to the Board for consideration as appropriate. The Company considers good corporate governance to be central to the effective and efficient management and operation of the Company, and the Board is directly responsible for developing the Company's approach to corporate governance issues. A discussion of the Company's governance practices within the context of NI 58-101 is set out below.

Statement of Corporate Governance Practices

Corporate governance relates to the activities of the board of directors of the Company (the "Board"), the members of which are elected by and are accountable to the shareholders, and takes into account the role of the individual members of management who are appointed by the Board and who are charged with the day-to-day management of the Company. The Board is committed to sound corporate governance practices which are both in the interest of its shareholders and contribute to effective and efficient decision making. National Policy 58-201 *Corporate Governance Guidelines* establishes corporate governance guidelines which apply to all public companies. The Company has reviewed its own corporate governance practices in light of these guidelines. In certain cases, the Company's practices comply with the guidelines, however, the Board considers that some of the guidelines are not suitable for the Company at its current stage of development and therefore these guidelines have not been adopted. National Instrument 58-101 *Disclosure of Corporate Governance Practices* mandates disclosure of corporate governance practices for Venture Issuers in Form 58-101F2, which disclosure is set out below.

Board of Directors

Structure and Compensation

The Board is currently composed of six (6) directors. All six of the proposed nominees for election as director at the Meeting are current directors of the Company.

Form 58-101F1 suggests that the board of directors of every listed corporation should be constituted with a majority of individuals who qualify as "independent" directors under National Instrument 52-110 ("NI 52-110"), which provides that a director is independent if he or she has no direct or indirect "material relationship" with the Company. "Material relationship" is defined as a relationship which could, in the view of the Company's board of directors, be reasonably expected to interfere with the exercise of a director's independent judgment. Of the proposed nominees, two (2), Fredy Salazar, President and Chief Executive Officer, Pablo Acosta, Chief Financial Officer, are "inside" or management directors and accordingly are considered not "independent". Nick DeMare, Etienne Walter, Merlin Marr-Johnson and Mary Gilzean are considered by the Board to be "independent", within the meaning of NI 52-110.

In assessing Form 58-101F2 and making the foregoing determinations, the circumstances of each director have been examined in relation to a number of factors.

The Board does not currently have a Chair. The independent directors exercise their responsibilities for independent oversight of management, and are provided with leadership through their position on the Board and ability to meet independently of management whenever deemed necessary.

The quantity and quality of the Board compensation is reviewed on an annual basis. At present, the Board is satisfied that the current Board compensation arrangements, which currently only includes incentive stock options, adequately reflect the responsibilities and risks involved in being an effective director of the Company. The number of options to be granted is determined by the Board as a whole, thereby providing the independent directors with significant input into compensation decisions. At this time the Company does not believe its size and limited scope of operations requires a formal compensation committee.

Directorships

As of the date of this information circular, certain directors of the Company are also serving as directors of other reporting issuers, details of which are as follows:

Director	Other Reporting Issuer
Fredy Salazar	N/A
Pablo Acosta	N/A
Nick DeMare	Auscan Resources Inc. (formerly American Helium Inc.), Cliffmont Resources Ltd., East West Petroleum Corp., Hannan Metals Ltd., Kingsmen Resources Ltd., Mirasol Resources Ltd., Rochester Resources Ltd., T2 Metals Corp. (formerly Aguila Copper Corp.), Tinka Resources Limited, Tribeca Resources Corporation (formerly Hansa Resources Limited), Whitewater Acquisition Corp.
Etienne Walter	N/A
Merlin Marr-Johnson	N/A
Mary Gilzean	N/A

Mandate of the Board

The mandate of the Board is to manage or supervise the management of the business and affairs of the Company and to act with a view to the best interests of the Company. In doing so, the Board oversees the management of the Company's affairs directly and through the Audit Committee. In fulfilling its mandate, the Board, among other matters, is responsible for reviewing and approving the Company's overall business strategies and its annual business plan, reviewing and approving the annual corporate budget and forecast, reviewing and approving significant capital investments outside the approved budget; reviewing major strategic initiatives to ensure that the Company's proposed actions accord with shareholder objectives; reviewing succession planning; assessing management's performance against approved business plans and industry standards; reviewing and approving the reports and other disclosure issued to shareholders; ensuring the effective operation of the Board; and safeguarding shareholders' equity interests through the optimum utilization of the Company's capital resources. The Board also takes responsibility for identifying the principal risks of the Company's business and for ensuring these risks are effectively monitored and mitigated to the extent reasonably practicable. At this stage of the Company's development, the Board does not believe it is necessary to adopt a written mandate, as sufficient guidance is found in the applicable corporate legislation and regulatory policies. However, as the Company grows, the Board will move to develop a formal written mandate.

In keeping with its overall responsibility for the stewardship of the Company, the Board is responsible for the integrity of the Company's internal control and management information systems and for the Company's policies respecting corporate disclosure and communications.

Each member of the Board understands that he is entitled to seek the advice of an independent expert if he reasonably considers it warranted under the circumstances.

The positions of President and Chief Executive Officer are combined. The Board believes the Company is well serviced and the independence of the Board from management is not compromised by the combined role. The Board does not, and does not consider it necessary to, have any formal structures or procedures in place to ensure that the Board can function independently of management. The Board believes that its current composition, in which only two (2) of six (6) are members of management, is sufficient to ensure that the Board can function independently of management.

Nomination and Assessment

The Board determines new nominees to the Board, although a formal process has not been adopted. The nominees are generally the result of recruitment efforts by the Board members, including both formal and informal discussions among Board members and the President and Chief Executive Officer. The Board monitors but does not formally assess the performance of individual Board members or committee members or their contributions.

The Board does not, at present, have a formal process in place for assessing the effectiveness of the Board as a whole, its committees or individual directors, but will consider implementing one in the future should circumstances warrant. Based on the Company's size, its stage of development and the limited number of individuals on the Board, the Board considers a formal assessment process to be inappropriate at this time. The Board plans to continue evaluating its own effectiveness on an ad hoc basis. The current size of the Board is such that the entire Board takes responsibility for selecting new directors and assessing current directors. Proposed directors' credentials are reviewed in advance of a Board Meeting with one or more members of the Board prior to the proposed director's nomination.

New directors are briefed on strategic plans, short, medium and long term corporate objectives, business risks and mitigation strategies, corporate governance guidelines and existing corporation policies. However, there is no formal orientation for new members of the Board, and this is considered to be appropriate, given the Company's size and current limited operations.

The skills and knowledge of the Board of Directors as a whole is such that no formal continuing education process is currently deemed required. The Board is comprised of individuals with varying backgrounds, who have, both collectively and individually, extensive experience in running and managing public companies in the natural resource sector. Board members are encouraged to communicate with management, auditors and technical consultants to keep themselves current with industry trends and developments and changes in legislation, with management's assistance. Board members have full access to the Company's records. Reference is made to the table under the heading "Election of Directors" in the Information Circular for a description of the current principal occupations of the Company's Board.

Expectations of Management and Ethical Business Conduct

The Board expects management to operate the business of the Company in a manner that enhances shareholder value and is consistent with the highest level of integrity. Management is expected to execute the Company's business plan and to meet performance goals and objectives. To date, the Board has not adopted a formal written Code of Business Conduct and Ethics. However, the current limited size of the Company's operations and the small number of officers and employees allow the independent members of the Board to monitor on an ongoing basis the activities of management and to ensure that the highest standard of ethical conduct is maintained. As the Company grows in size and scope, the Board anticipates that it will formulate and implement a formal Code of Business Conduct and Ethics.

Committee Responsibilities and Activities

Committees of the Board are an integral part of the Company's governance structure. At the present time, the only standing committee is the Audit Committee. Disclosure with respect to the Audit Committee, as required by NI 52-110 – Audit Committee, is contained in Schedule "A" to this Information Circular. As the Company grows, and its operations and management structure become more complex, the Board will likely find it appropriate to constitute formal standing committees, such as a Corporate Governance Committee, Compensation Committee and Nominating Committee, and to ensure that such committees are governed by written charters and are composed of at least a majority of independent directors.

PARTICULARS OF OTHER MATTERS TO BE ACTED UPON

A. Approval of Option Plan Amendments

Shareholders at the annual meeting held on October 14, 2021 approved the re-implementation of the Company's "rolling" Option Plan, which provides that the maximum number of common shares that may be reserved for issuance pursuant to such Option Plan will not exceed 10% of the issued shares of the Company at the time of the stock option grant. For a description of the Option Plan, see "Stock Option Plans and Other Incentive Plans". In accordance with the policies of the Exchange, a rolling plan requires the approval of the shareholders of the Company on an annual basis.

On November 24, 2021, the Exchange adopted a new policy 4.4 governing security based compensation ("New Policy 4.4"). The changes to the policy generally relate to the expansion of the policy to cover a number of types of security based compensation in addition to stock options. In order to comply with the new Policy 4.4, the Company has determined to amend its Option Plan.

The Board has approved, subject to receipt of shareholder and Exchange approval, certain amendments to the Option Plan. The principal amendments to the Option Plan are summarized below. The amendments (the "Amendments") to the Option Plan (the Option Plan, as amended, being referred to as the "Amended Option Plan") will then be submitted to shareholders at the Meeting for approval. A copy of the Amended Option Plan, showing the changes from the Option Plan, is attached as Schedule "B" to this Information Circular.

The Amendments to the Option Plan include:

- (a) updating certain defined terms, such as the defined terms for "Black-Out", "Consultant", and "Employee" to reflect the definitions used by New Policy 4.4;
- (b) the addition of new defined terms, such as the defined terms for "Director", "Disinterested Shareholder Approval", "Investor Relations Activities", "Investor Relations Service Provider", "Management Company Employee", "Material Change", "Material Fact", "Material Information", "Officer" and "Securities Laws";
- (c) the addition of a new subsection 2.3(b) which provides that the maximum aggregate number of Shares that are issuable to insiders (as a group) pursuant to the exercise of Options and pursuant to any other security based compensation arrangement must not exceed 10% of the Outstanding Issue at any point in time, unless the Company has obtained Disinterested Shareholder Approval;

- (d) the addition of a new subsection 2.3(f) which provides that Investor Relations Service Providers may not receive any security based compensation other than Options;
- (e) revisions to the provisions in sections 2.3(c) (setting an annual 10% limit of the maximum aggregate number of Shares that may be issued to insiders in any 12 month period pursuant to the exercise of Options and pursuant to any other security based compensation arrangement, 2.3(e) (clarifying the annual option limit for consultants), and 2.3(g) (clarifying the vesting requirements for Options granted to Investor Relations Service Providers) in order to better comply with New Policy 4.4;
- (f) revisions to the provisions in sections 3.2(c) to clarify that the number of Common Shares which will be available for purchase pursuant to Options granted pursuant to this Plan, together with the number of Shares issuable at any time under all other security based compensation arrangements, will not exceed 10% of the number of Common Shares which are issued and outstanding on the particular date of grant of Options;
- (f) revisions to the provision in section 5.2, Extension of Expiry Date of Stock Options Due to a Black-Out of the Option Plan to better comply with New Policy 4.4;
- (g) revisions to the provision in section 5.6 of the Option Plan to include that there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the Exchange, in accordance with the New Policy 4.4;
- (h) revisions to the provision in section 7.1 of the Option Plan to include that Options held by Investor Relations Service Providers may not be exercised on a "net exercise" basis, in accordance with the New Policy 4.4;
- (i) revisions to the provision in section 8.4(h) of the Option Plan to include that there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the Exchange, in accordance with the New Policy 4.4;
- (j) revisions to the provision in section 9.2 of the Option Plan to include that Disinterested Shareholder Approval is required for any reduction in the exercise price of an Option, or the extension of the term of an Option, if the Option holder is an insider of the Company at the time of the proposed amendment, in accordance with the New Policy 4.4;
- (f) the addition of a new subsection 11.3(h) which provides that any adjustment, other than in connection with a share consolidation or share split, to Options granted or issued under the Option Plan are subject to the prior acceptance of the Exchange, including adjustments related to an amalgamation, merger, arrangement, reorganization, spin-off, dividend or recapitalization, in accordance with the New Policy 4.4; and
- (g) revisions to the provision in section 11.5 of the Option Plan to include that there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the Exchange, in accordance with the New Policy 4.4.

In addition, certain other amendments of a housekeeping nature were made.

The existing options which are outstanding under the Option Plan will be incorporated into the Amended Option Plan and will be governed by the Amended Option Plan.

Copies of the Option Plan and Amended Option Plan are available for viewing up to the date of the Meeting at the Company's offices at Suite 1305, 1090 West Georgia Street, Vancouver, British Columbia, V6E 3V7 during normal business hours and at the Meeting. In addition, a copy of the Option Plan and Amended Option Plan will be mailed, free of charge, to any holder of Common Shares who requests a copy, in writing, from the Secretary of the Company. Any such requests should be mailed to the Company, at its head office, to the attention of the Secretary.

Option Plan Amendment Resolution

At the Meeting, Shareholders will be asked to consider and, if thought appropriate, to pass the following ordinary resolution, in substantially the following form, approving the Amendments and the Amended Option Plan (the "Option Plan Amendment Resolution").

"RESOLVED that:

- 1. the Stock Option Plan, being a "rolling" stock option plan, of Salazar Resources Limited, as amended by the board of directors and substantially in the form described in the information circular dated January 12, 2023 and presented to the shareholders (the "Amended Option Plan"), be and is hereby approved;
- 2. the number of Common Shares reserved for issuance under the Option Plan, as amended, shall be no more than 10% of the Company's issued and outstanding share capital at the time of any stock option grant;
- 3. the board of directors be authorized on behalf of the Company to make any further amendments to the Amended Option Plan as may be required by regulatory authorities, without further approval of the shareholders of the Company, in order to ensure adoption of the Amended Option Plan; and
- 4. the approval of the Amended Option Plan by the board of directors of the Company is hereby ratified and confirmed any one director or officer of the Company is hereby authorized and directed for and in the name of and on behalf of the Company to execute or cause to be executed, whether under corporate seal of the Company or otherwise, and to deliver or cause to be delivered all such documents, and to do or cause to be done all such acts and things as in the opinion of such director or officer may be necessary or desirable in connection with the foregoing."

The Board recommends that Shareholders vote in favour of the above Option Plan Amendment Resolution. In the absence of a contrary instruction, the persons named in the enclosed form of proxy intend to vote in favour of the Option Plan Amendment Resolution.

To be effective, the Option Plan Amendment Resolution must be approved by at least a majority of the votes cast thereon at the Meeting.

B. Amendment of Restricted Share Unit Plan

The Company's Restricted Share Unit Plan ("RSU Plan") was first approved by the shareholders on September 25, 2020. As of the Record Date, a total of 863,000 RSUs had been awarded under the RSU Plan and there are 137,000 RSUs available for award under the current RSU Plan. The Board continues to believe that it is in the Company's best interest to have the RSU Plan to assist the Company in attracting, retaining and motivating employees and directors and to more fully align their economic interests with those of the shareholders. See "Director and Named Executive Officer Compensation — Stock Option Plans and Other Incentive Plans — Restricted Share Unit Plan" for a summary of the current provisions of the RSU Plan.

As noted above, on November 24, 2021, the Exchange adopted New Policy 4.4 which requires the Company to amend the RSU Plan.

On December 6, 2022 the Board approved, subject to shareholder and Exchange approval, certain amendments to the RSU Plan, to bring it into compliance with New Policy 4.4 and to also increase the total number of RSUs available for award under the RSU Plan from 1,000,000 RSUs to 2,000,000 RSUs. The principal amendments to the RSU Plan (the "RSU Amendments") are summarized below. Such amendments to the RSU Plan will then be submitted to shareholders at the Meeting for approval. A copy of the RSU Plan with the RSU Amendments (the "Amended RSU Plan") showing the proposed changes to the RSU Plan, is attached as Schedule "C" to this Information Circular.

The RSU Amendments to the RSU Plan include:

- (a) the addition of new defined terms, such as the defined terms for "Black-Out", "Consultant", "Director", "Disinterested Shareholder Approval", "Employee", "Investor Relations Service Provider", "Management Company Employee", "Material Change", "Material Fact", "Material Information", "Officer", and "Securities Laws", to reflect the definitions used by New Policy 4.4;
- (b) the addition of a new subsection 2.6(b) which provides that Disinterested Shareholder Approval is required for any extension of the term of an RSU, if the RSU holder is an Insider of the Company at the time of the proposed amendment;
- (c) the addition of a new paragraph under subsection 4.1 which provides that Investor Relations Service Providers may not receive any RSUs pursuant to the RSU Plan;
- (d) the revision to the provision in subsection 4.4(a) of the RSU Plan to include that the maximum number of Shares issuable at any time pursuant to RSUs under the RSU Plan shall be limited to 2,000,000;
- (e) the revisions to the provisions in subsections 4.4(b), (c), (d), and (e) of the RSU Plan to clarify that the maximum aggregate number of Shares that may be reserved for issuance to any Insiders (as a group), includes all Security Based Compensation Arrangements, including the Company's stock option plan, with the calculation to occur as at the date of grant or issue;
- (f) the addition of a new paragraph under section 4.5 of the RSU Plan to clarify that the automatic extension is available to all eligible RSU holders under the RSU Plan under the same terms and conditions, provided that such automatic extension will not be permitted where the RSU holder or the Company is subject to a cease trade order (or similar order under Securities Laws) in respect of the Company's securities; and
- (g) the addition of a new paragraph under section 5.1 of the RSU Plan to clarify that any adjustment, other than in connection with a share consolidation or share split, to RSUs granted or issued under the RSU Plan are subject to the prior acceptance of the Exchange, including adjustments related to an amalgamation, merger, arrangement, reorganization, spin-off, dividend or recapitalization.

In addition, certain other amendments of a housekeeping nature were made.

The Company is seeking the approval of disinterested Shareholders at the Meeting to pass the following ordinary resolution, in substantially the following form, approving the RSU Amendments and the Amended RSU Plan (the "RSU Plan Amendment Resolution"). Such approval is required under the rules and regulations of the TSXV. If the Shareholders do not approve the RSU Plan Amendment Resolution, the existing maximum number of RSUs available for award under the RSU Plan will remain in place.

RSU Plan Amendment Resolution

In accordance with the policies of the Exchange, the RSU Plan Amendment Resolution must be passed by a majority of the votes cast on the ordinary resolution by disinterred shareholders at the Meeting. If the Shareholders do not approve the RSU Amendments, the existing maximum number of RSUs available for award under the RSU Plan will remain in place.

To be passed, the RSU Plan Amendment Resolution must be approved by a majority of votes cast by disinterested Shareholders at the Meeting, present in person or by Proxy. A total of 37,212,367 Common Shares held by directors and executive officers of the Company will be excluded. The Board recommends that disinterested Shareholders vote in favour of the proposed resolution. In the absence of contrary instructions, the persons named in the accompanying form of proxy intend to vote the Common Shares represented thereby FOR the RSU Plan Amendment Resolution.

The RSU Plan Amendment Resolution, which must be approved by a majority of disinterested Shareholders at the Meeting, is as follows:

"RESOLVED as an ordinary resolution of disinterested shareholders that:

- 1. the Restricted Share Unit Plan of Salazar Resources Limited, as amended by the board of directors and substantially in the form described in the information circular dated January 12, 2023 and presented to the shareholders (the "Amended RSU Plan"), be and is hereby approved;
- 2. the increase in the number of restricted share units available for award thereunder to 2,000,000 restricted share units, be and is hereby approved;
- 3. the Company is hereby authorized and directed to issue such Common Shares pursuant to the Amended RSU Plan as fully paid and non-assessable Common Shares of the Company;
- 4. the board of directors of the Company be authorized to make any changes to the Amended RSU Plan, as may be required or permitted by the TSX Venture Exchange; and
- 5. any one director or officer of the Company is authorized and directed, on behalf of the Company, to take all necessary steps and proceedings and to execute, deliver and file any and all declarations, agreements, documents and other instruments and do all such other acts and things that may be necessary or desirable to give effect to this ordinary resolution."

The Board recommends that Shareholders vote in favour of the above RSU Plan Amendment Resolution. In the absence of a contrary instruction, the persons named in the enclosed form of proxy intend to vote in favour of the RSU Plan Amendment Resolution.

To be effective, the RSU Plan Amendment Resolution must be approved by at least a majority of the disinterested votes cast thereon at the Meeting.

ANY OTHER MATTERS

Management of the Company knows of no matters to come before the Meeting other than those referred to in the Notice of Meeting accompanying this Information Circular. However, if any other matters properly come before the Meeting, it is the intention of the persons named in the form of proxy accompanying this Information Circular to vote the same in accordance with their best judgment of such matters.

ADDITIONAL INFORMATION

Additional information regarding the Company and its business activities is available on the Company's profile on SEDAR website located at www.sedar.com. The Company's financial information is provided in the Company's audited comparative financial statements and related management discussion and analysis for its most recently completed financial year and may be viewed on the SEDAR website at the website noted above. Shareholders of the Company may request copies of the Company's financial statements and related management discussion and analysis by contacting the Company at Salazar Resources Limited, #1305 – 1090 W. Georgia Street, Vancouver, BC, V6E 3V7, Tel: (604) 685-9316 | Fax: (604) 683-1585.

Schedule "A"

SALAZAR RESOURCES LIMITED

CHARTER OF THE AUDIT COMMITTEE OF THE BOARD OF DIRECTORS

Mandate

The primary function of the audit committee (the "Committee") is to assist the Board of Directors in fulfilling its financial oversight responsibilities by reviewing the financial reports and other financial information provided by the Company to regulatory authorities and shareholders, the Company's systems of internal controls regarding finance and accounting and the Company's auditing, accounting and financial reporting processes. The Committee's primary duties and responsibilities are to:

- (i) Serve as an independent and objective party to monitor the Company's financial reporting and internal control system and review the Company's financial statements.
- (ii) Review and appraise the performance of the Company's external auditors (the "Auditor").
- (iii) Provide an open avenue of communication among the Company's auditors, financial and senior management and the Board of Directors.

Composition

The Committee shall be comprised of at least three directors. Each member must be a director of the Company. A majority of the members of the Committee shall not be officers or employees of the Company or of an affiliate of the Company. At least one member of the Committee shall be financially literate. All members of the Committee that are not financially literate will work towards becoming financially literate to obtain a working familiarity with basic finance and accounting practices. For the purposes of the Audit Committee Charter, the term "financially literate" means the ability to read and understand a set of financial statements that present a breadth and level of complexity of accounting issues that are generally comparable to the breadth and complexity of the issues that can reasonably be expected to be raised by the Company's financial statements.

The members of the Committee shall be appointed by the Board of Directors at its first meeting following the annual shareholders' meeting. Unless a Chair is elected by the full Board of Directors, the members of the Committee may designate a Chair by a majority vote of the full Committee membership. The Chair shall be financially literate.

Meetings

Meetings of the Committee shall be scheduled to take place at regular intervals. Unless all members are present and waive notice, or those absent waive notice before or after a meeting, the Chairman will give Committee members 24 hours' advance notice of each meeting and the matters to be discussed at it. Notice may be given personally, by telephone, facsimile or email.

The Auditor shall be given reasonable notice of, and be entitled to attend and speak at, each meeting of the Committee concerning the Company's annual financial statements and, if the Committee feels it is necessary or appropriate, at any other meeting. On request by the Auditor, the Chair shall call a meeting of the Committee to consider any matter that the Auditor believes should be brought to the attention of the Committee, the Board of Directors or the shareholders of the Company.

At each meeting of the Committee, a quorum shall consist of a majority of members that are not officers or employees of the Company or of an affiliate of the Company. A member may participate in a meeting of the Committee in person or by telephone if all members participating in the meeting, whether in person or by telephone or other communications medium, are able to communicate with each other. A member may participate in a meeting of the Committee by a communications medium other than telephone if all members participating in the meeting, whether in person or by

telephone or other communications medium, are able to communicate with each other and if all members who wish to participate in the meeting agree to such participation.

As part of its goal to foster open communication, the Committee may periodically meet separately with each of management and the Auditor to discuss any matters that the Committee or any of these groups believes would be appropriate to discuss privately. In addition, the Committee should meet with the Auditor and management annually to review the Company's financial statements.

The Committee may invite to its meetings any director, any manager of the Company, and any other person whom it deems appropriate to consult in order to carry out its responsibilities. The Committee may also exclude from its meetings any person it deems appropriate to exclude in order to carry out its responsibilities.

Responsibilities and Duties

To fulfill its responsibilities and duties, the Committee shall:

Documents/Reports Review

- (a) Review the Company's financial statements, including any certification, report, opinion, or review rendered by the Auditor, MD&A and any annual and interim earnings press releases before the Company publicly discloses this information.
- (b) Review and satisfy itself that adequate procedures are in place for the review of the Company's public disclosure of financial information extracted or derived from its financial statements, other than disclosure described in the previous paragraph, and periodically assess the adequacy of those procedures.

External Auditors

- (a) Be directly responsible for overseeing the work by the Auditor (including resolution of disagreements between management and the Auditor regarding financial reporting) engaged for the purpose of preparing or issuing an audit report or performing other audit or review services for the Company.
- (b) Require the Auditor to report directly to the Committee.
- (c) Review annually, the performance of the Auditor who shall be ultimately accountable to the Board of Directors and the Committee as representatives of the shareholders of the Company.
- (d) Review and discuss with the Auditor any disclosed relationships or services that may impact the objectivity and independence of the Auditor.
- (e) Take, or recommend that the Board of Directors take, appropriate action to oversee the independence of the Auditor.
- (f) Recommend to the Board of Directors the external auditor to be nominated at the annual general meeting for appointment as the Auditor for the ensuing year and the compensation for the Auditors, or, if applicable, the replacement of the Auditor.
- (g) Review and approve the Company's hiring policies regarding partners, employees and former partners and employees of the Auditor and former independent external auditor of the Company.
- (h) Be directly responsible for the oversight of the work by the Auditor (including resolution of disagreements between management and the Auditor regarding financial reporting) for the purpose of preparing or issuing an audit report or related work.
- (i) Review with management and the Auditor the audit plan for the year end financial statements.

- (j) Review and pre-approve all audit and audit related services and the fees and other compensation related thereto, and any non-audit services, provided by the Auditor. The pre-approval requirement is waived with respect to the provision of non-audit services if:
 - (i) the aggregate amount of all such non-audit services that were not pre-approved is reasonably expected to constitute not more than 5% of the total amount of fees paid by the Company and its subsidiary entities to the Auditor during the fiscal year in which the non-audit services are provided;
 - (ii) such services were not recognized by the Company at the time of the engagement to be non-audit services; and
 - (iii) such services are promptly brought to the attention of the Committee and approved, prior to the completion of the audit, by the Committee or by one or more members of the Committee to whom authority to grant such approvals has been delegated by the Committee.

The Committee may delegate to one or more independent members of the Committee the authority to pre-approve non-audit services in satisfaction of the pre-approval requirement set out in this section provided the pre-approval of non-audit services by any member to whom authority has been delegated must be presented to the Committee at its first scheduled meeting following such pre-approval.

Financial Reporting Processes

- (a) In consultation with the Auditor, review with management the integrity of the Company's financial reporting process, both internal and external.
- (b) Consider the Auditor's judgments about the quality and appropriateness of the Company's accounting principles as applied in its financial reporting.
- (c) Consider and approve, if appropriate, changes to the Company's auditing and accounting principles and practices as suggested by the Auditor and management.
- (d) Review significant judgments made by management in the preparation of financial statements and the view of the Auditor as to the appropriateness of such judgments.
- (e) Following completion of the annual audit, review separately with management and the Auditor any significant difficulties encountered during the course of the audit, including any restrictions on the scope of work or access to required information.
- (f) Review any significant disagreement among management and the Auditor in connection with the preparation of the financial statements.
- (g) Review with the Auditor and management the extent to which changes and improvements in financial or accounting practices have been implemented.
- (h) Discuss with the Auditor the Auditor's perception of the Company's financial and accounting personnel, any material recommendations which the Auditor may have, the level of cooperation which the Auditor received during the course of their review and the adequacy of their access to records, data and other requested information.
- (i) Review any complaints or concerns about any questionable accounting, internal accounting controls or auditing matters.
- (j) Establish procedures for:
 - (i) the receipt, retention and treatment of complaints received by the Company regarding accounting, internal accounting controls, or auditing matters; and

(ii) the confidential, anonymous submission by the employees of the Company of concerns regarding questionable accounting or auditing matters.

Other

- (a) Perform such other duties as may be assigned to it by the Board of Directors from time to time or as may be required by applicable regulatory authorities or legislation.
- (b) Report regularly and on a timely basis to the Board of Directors on matters coming before the Committee.
- (c) Review and reassess the adequacy of this Charter annually and recommend any proposed changes to the Board of Directors for approval.

Authority

The Committee is authorized to:

- (a) seek any information it requires from any employee of the Company in order to perform its duties;
- (b) engage, at the Company's expense, independent legal counsellors or other professional advisors on any matter within the scope of the role and duties of the Committee under this Charter;
- (c) set and pay the compensation for any advisors engaged by the Committee; and
- (d) communicate directly with the internal and external auditors of the Company.

This Charter supersedes and replaces all prior charters and other terms of reference pertaining to the Committee.

Schedule "B"

SALAZAR RESOURCES LIMITED

STOCK OPTION PLAN

Effective Date: September 25, 2014₂
as amended , 2023

Approved by the Board of Directors on August 25, 2014, as amended •, 2023.

TABLE OF CONTENTS

		<u>Page</u>
CECT	ION 4 DEPINITIONS AND INTERDRET ATION	1
SEC 1 .1	ION 1 DEFINITIONS AND INTERPRETATION Definitions	
1.1	Choice of Law	
1.2	Headings	-
	ION 2 PURPOSE AND PARTICIPATION	
2.1	Purpose of Plan	
2.1	Participation in Plan	
2.3	Limits on Option Grants	
2.3	Notification of Grant	
2.5	Copy of Plan	
2.6	Limitation on Service	-
2.7	No Obligation to Exercise	
2.8	Agreement	
2.9	Notice	
	Representation to TSXV	_
	ION 3 NUMBER OF SHARES UNDER PLAN	
3.1	Board to Approve Issuance of Shares	-
_	Number of Shares	
	Fractional Shares	
	ION 4 GRANT OF OPTIONS	
4.1	Grant of Options	
4.2	Record of Option Grants	
4.3	Effect of Plan	
_	ION 5 TERMS AND CONDITIONS OF OPTIONS	
5.1	Exercise Period of Option	
5.2	Extension of Expiry Date of Stock Options Due to a Black-Out	
5.3	Number of Shares Under Option	
5.4	Exercise Price of Option	
5.5	Termination of Option	
5.6	Vesting of Option and Acceleration	
5.7	Additional Terms	
SECT	ION 6 TRANSFERABILITY OF OPTIONS	
6.1	Non-transferable	
6.2	Death of Option Holder	
6.3	Disability of Option Holder	<u>10</u> 11
6.4	Disability and Death of Option Holder	
6.5	Vesting	<u>10</u> 11
6.6	Deemed Non-Interruption of Engagement	<u>10</u> 11
SECT	ION 7 EXERCISE OF OPTION	
7.1	Exercise of Option	
7.2	Issue of Share Certificates	
7.3	No Rights as Shareholder	<u>11</u> 12
SECT	ION 8 ADMINISTRATION	<u>11<u>12</u></u>
8.1	Board or Committee	<u>41</u> 12
8.2	Appointment of Committee	<u>41<u>12</u></u>

8.3	Quorum and Voting	. 11 12	
8.4	Powers of Committee	. 12 13	į
8.5	Administration by Committee	. 12 13	į
	Interpretation		
	ON 9 APPROVALS AND AMENDMENT		
	Shareholder Approval of Plan		
	Amendment of Option or Plan		
	ON 10 CONDITIONS PRECEDENT TO ISSUANCE OF OPTIONS AND SHARES		
	Compliance with Laws		
	Obligation to Obtain Regulatory Approvals		-
	Inability to Obtain Regulatory Approvals		
	Withholding Tax Requirements		
	ON 11 ADJUSTMENTS AND TERMINATION		
	Termination of Plan		
	No Grant During Suspension of Plan		
	Alteration in Capital Structure		
	Triggering Events		
	Notice of Termination by Triggering Event		
	Determinations to be Made By Committee		-
	Options Granted to U.S. Residents or Citizens		

STOCK OPTION PLAN

SECTION 1 DEFINITIONS AND INTERPRETATION

1.1 Definitions

As used herein, unless there is something in the subject matter or context inconsistent therewith, the following terms shall have the meanings set forth below:

- (a) "Administrator" means such Executive or Employee of the Company as may be designated as Administrator by the Committee from time to time, if any.
- (b) "Associate" means, where used to indicate a relationship with any person:
 - (i) any relative, including the spouse of that person or a relative of that person's spouse, where the relative has the same home as the person;
 - (ii) any partner, other than a limited partner, of that person;
 - (iii) any trust or estate in which such person has a substantial beneficial interest or as to which such person serves as trustee or in a similar capacity; and
 - (iv) any corporation of which such person beneficially owns or controls, directly or indirectly, voting securities carrying more than 10% of the voting rights attached to all outstanding voting securities of the corporation.
- (c) "Black-Out" means a restriction imposed by the Company <u>pursuant to the Company's internal policies</u> on all or any of its <u>directors</u>, <u>officers</u>, <u>employees</u>, <u>insiders</u> <u>Directors</u>, <u>Officers</u>, <u>Employees</u>, <u>Insiders</u> or persons in a special relationship <u>as a result of the bona fide existence of undisclosed Material Information</u> whereby they are to refrain from trading in the Company's securities until the restriction has been lifted by the Company. Such restriction must expire following the general disclosure of the undisclosed Material Information.
- (d) "Board" means the board of directors of the Company.
- (e) "Change of Control" means an occurrence when either:
 - (i) a Person or Entity, other than the current "control person" of the Company (as that term is defined in the *Securities Act*), becomes a "control person" of the Company; or
 - (ii) a majority of the directors elected at any annual or extraordinary general meeting of shareholders of the Company are not individuals nominated by the Company's then-incumbent Board.
- (f) "Committee" means a committee of the Board appointed in accordance with this Plan or if no such committee is appointed, the Board itself.
- (g) "Company" means Salazar Resources Limited
- (h) "Consultant" means, in relation to the Company, an individual (other than a Director, Officer or Employee of the Company) who:

- (i) is engaged to provide, on an ongoing *bona fide* basis, consulting, technical, management or other services to the Company or any Subsidiary other than services provided in relation to a "distribution" (as that term is described in the *Securities Act*);
- (ii) provides the services under a written contract between the Company or any Subsidiary and the individual or a Consultant Entity (as defined in clause (h)(viv) below); and
- (iii) in the reasonable opinion of the Company, spends or will spend a significant amount of time and attention on the affairs and business of the Company or any Subsidiary; and
- (iv) has a relationship with the Company or any Subsidiary that enables the individual to be knowledgeable about the business and affairs of the Company or is otherwise permitted by applicable Regulatory Rules to be granted Options as a Consultant or as an equivalent thereof,

and includes:

- (iv) a corporation of which the individual is an employee or shareholder or a partnership of which the individual is an employee or partner (a "Consultant Entity"); or
- (vi) an RRSP or RRIF established by or for the individual under which he or she is the beneficiary.
- (i) "Director" means a director (as defined under Securities Laws) of the Company or any of its Subsidiaries.
- (i) "Disability" means a medically determinable physical or mental impairment expected to result in death or to last for a continuous period of not less than 12 months, and which causes an individual to be unable to engage in any substantial gainful activity, or any other condition of impairment that the Committee, acting reasonably, determines constitutes a disability.
- (k) "Disinterested Shareholder Approval" has the meaning ascribed thereto by the TSXV in "Policy 4.4 Security Based Compensation" of the TSXV's Corporate Finance Manual.
- (1) (i) "Employee" means:
 - (i) an individual who works full-time or part-time for the Company or <u>any of its</u> Subsidiary and such other individual as may, from time to time, be permitted by applicable Regulatory Rules to be granted Options as an employee or as an equivalent thereto; or
 - (ii) an individual who works for the Company or <u>anyits</u> Subsidiary either full-time or on a continuing and regular basis for a minimum amount of time per week providing services normally provided by an employee and who is subject to the same control and direction by the Company or <u>anyits</u> Subsidiary over the details and methods of work as an employee of the Company or <u>anyof the</u> Subsidiary, <u>as the case may be</u>, but for whom income tax deductions are not made at source,

and includes:

- (iii) a corporation wholly-owned by such individual; and
- (iv) any RRSP or RRIF established by or for such individual under which he or she is the beneficiary.
- (m) (k)—"Executive" means an individual who is a director Director or officer Officer of the Company or a Subsidiary, and includes:
 - (i) a corporation wholly-owned by such individual; and

- (ii) any RRSP or RRIF established by or for such individual under which he or she is the beneficiary.
- (n) "Exercise Notice" means the written notice of the exercise of an Option, in the form set out as Schedule "B" hereto, duly executed by the Option Holder.
- (o) (m) "Exercise Period" means the period during which a particular Option may be exercised and is the period from and including the Grant Date through to and including the Expiry Time on the Expiry Date provided, however, that no Option can be exercised unless and until all necessary Regulatory Approvals have been obtained.
- (p) (n) "Exercise Price" means the price at which an Option is exercisable as determined in accordance with section 5.4.
- (g) (e) "Expiry Date" means the date the Option expires as set out in the Option Certificate or as otherwise determined in accordance with sections 5.2, 5.5, 6.2, 6.3, 6.4 or 11.4.
- (r) (p) "Expiry Time" means the time the Option expires on the Expiry Date, which is 5:00 p.m. local time in Vancouver, British Columbia on the Expiry Date.
- (s) (q)-"Grant Date" means the date on which the Committee grants a particular Option, which is the date the Option comes into effect provided however that no Option can be exercised unless and until all necessary Regulatory Approvals have been obtained.
- (t) (r) "Insider" means an insider as that term is defined in the Securities Act;.
- (u) "Investor Relations Activities" has the meaning ascribed thereto by the TSXV in Policy 1.1 of the TSXV's Corporate Finance Manual.
- (v) "Investor Relations Service Provider" means any Consultant that performs Investor Relations Activities and any Director, Officer, Employee or Management Company Employee whose role and duties primarily consist of Investor Relations Activities.
- (w) "Management Company Employee" means an individual employed by a company providing management services to the Company, which services are required for the ongoing successful operation of the business enterprise of the Company.
- (x) (s) "Market Value" means the market value of the Shares as determined in accordance with section 5.4.
- (y) "Material Change" means the definition prescribed by applicable Securities Laws.
- (z) "Material Fact" means the definition prescribed by applicable Securities Laws.
- (aa) "Material Information" means a Material Fact and/or Material Change as defined by applicable Securities

 <u>Laws and TSXV Policy.</u>
- (bb) "Officer" means an officer (as defined under Securities Laws) of the Company or of any of its Subsidiaries.
- (cc) (t) "Option" means an incentive share purchase option granted pursuant to this Plan entitling the Option Holder to purchase Shares of the Company.
- (dd) (u)-"Option Certificate" means the certificate, in substantially the form set out as Schedule "A" hereto, evidencing the Option.

- (ee) (v) "Option Holder" means a Person or Entity who holds an unexercised and unexpired Option or, where applicable, the Personal Representative of such person.
- (m) "Outstanding Issue" means the number of Shares that are outstanding (on a non-diluted basis) immediately prior to the Share issuance or grant of Option in question.
- (gg) (x) "Person or Entity" means an individual, natural person, corporation, government or political subdivision or agency of a government, and where two or more persons act as a partnership, limited partnership, syndicate or other group for the purpose of acquiring, holding or disposing of securities of an issuer, such partnership, limited partnership, syndicate or group shall be deemed to be a Person or Entity.
- (hh) (y) "Personal Representative" means:
 - (i) in the case of a deceased Option Holder, the executor or administrator of the deceased duly appointed by a court or public authority having jurisdiction to do so; and
 - (ii) in the case of an Option Holder who for any reason is unable to manage his or her affairs, the person entitled by law to act on behalf of such Option Holder.
- (ii) (z) "Plan" means this stock option plan as from time to time amended.
- (ii) (aa) "Regulatory Approvals" means any necessary approvals of the Regulatory Authorities as may be required from time to time for the implementation, operation or amendment of this Plan or for the Options granted from time to time hereunder.
- (kk) "Regulatory Authorities" means all organized trading facilities on which the Shares are listed, and all securities commissions or similar securities regulatory bodies having jurisdiction over the Company, this Plan or the Options granted from time to time hereunder.
- (II) (ee)—"Regulatory Rules" means all corporate and securities laws, regulations, rules, policies, notices, instruments and other orders of any kind whatsoever which may, from time to time, apply to the implementation, operation or amendment of this Plan or the Options granted from time to time hereunder including, without limitation, those of the applicable Regulatory Authorities.
- (mm) (dd) "Securities Act" means the Securities Act (British Columbia), RSBC 1996, c.418 as from time to time amended.
- (nn) "Securities Laws" means securities legislation, securities regulation and securities rules, as amended, and the policies, notices, instruments and blanket orders in force from time to time that are applicable to the Company.
- (oo) (ee) "Share" or "Shares" means, as the case may be, one or more common shares without par value in the capital stockauthorized share structure of the Company.
- (pp) (ff) "Subsidiary" means a wholly-owned or controlled subsidiary corporation of the Company.
- (qq) (gg) "Triggering Event" means:
 - (i) the proposed dissolution, liquidation or wind-up of the Company;
 - (ii) a proposed merger, amalgamation, arrangement or reorganization of the Company with one or more corporations as a result of which, immediately following such event, the shareholders of the Company as a group, as they were immediately prior to such event, are expected to hold less than a majority of the outstanding capital stock of the surviving corporation;

- (iii) the proposed acquisition of all or substantially all of the issued and outstanding shares of the Company by one or more Persons or Entities;
- (iv) a proposed Change of Control of the Company;
- (v) the proposed sale or other disposition of all or substantially all of the assets of the Company; or
- (vi) a proposed material alteration of the capital structure of the Company which, in the opinion of the Committee, is of such a nature that it is not practical or feasible to make adjustments to this Plan or to the Options granted hereunder to permit the Plan and Options granted hereunder to stay in effect.
- (rr) (hh) "TSXV" means the TSX Venture Exchange Inc.
- (ss) (ii) "Vest" or "Vesting" means that a portion of the Option granted to the Option Holder which is available to be exercised by the Option Holder at any time and from time to time.

1.2 Choice of Law

The Plan is established under, and the provisions of the Plan shall be subject to and interpreted and construed in accordance with, the laws of the Province of British Columbia. The Company and each Option Holder hereby attorn to the jurisdiction of the Courts of British Columbia.

1.3 Headings

The headings used herein are for convenience only and are not to affect the interpretation of the Plan.

SECTION 2 PURPOSE AND PARTICIPATION

2.1 Purpose of Plan

The purpose of the Plan is to provide the Company with a share-related mechanism to attract, retain and motivate qualified Executives, Employees and Consultants, to incent such individuals to contribute toward the long term goals of the Company, and to encourage such individuals to acquire Shares of the Company as long term investments.

2.2 Participation in Plan

The Committee shall, from time to time and in its sole discretion, determine those Executives, Employees and Consultants, if any, to whom Options are to be granted.

2.3 Limits on Option Grants

If the Company is listed on TSXV, the following limitations shall apply to the Plan and all Options thereunder so long as such limitations are required by the TSXV:

- (a) the maximum number of Options which may be granted to any one Option Holder under the Plan within any 12 month period shall be 5% of the Outstanding Issue (unless the Company has obtained disinterested shareholder approval as required by the TSXV);
- (b) the maximum aggregate number of Shares that are issuable to Insiders (as a group) pursuant to the exercise of Options and pursuant to any other security based compensation arrangement must not exceed 10% of the Outstanding Issue at any point in time, unless the Company has obtained Disinterested Shareholder Approval;

- (c) (b) the maximum aggregate number of Options which may be Shares that are issuable pursuant to the exercise of Options and pursuant to any other security based compensation arrangement granted to Insiders within or issued in any 12 month period to Insiders (as a group) must not exceed 10% of the Outstanding Issue calculated as at the date any Options are granted or issued to any Insider (including any Options which are granted and exercised within that 12 month period), unless the Company has obtained disinterested shareholder approval as required by the TSXV) Disinterested Shareholder Approval;
- (d) (e) with respect to section 5.1, the Expiry Date of an Option shall be no later than the tenth anniversary of the Grant Date of such Option;
- (e) (d) the maximum <u>aggregate</u> number of Options which may be granted to any one Consultant within any 12 month period must not exceed 2% of the Outstanding Issue, <u>calculated as at the date an Option is granted or issued to the Consultant</u>; and
- (f) Investor Relations Service Providers may not receive any security based compensation other than Options;
- (g) (e)—the maximum <u>aggregate</u> number of Options which may be granted <u>withinin</u> any 12 month period to <u>Employees or Consultants engaged in investor relations activities</u> all <u>Investor Relations Service Providers</u> must not exceed 2% of the Outstanding Issue, <u>calculated as at the date any Option is granted to any such Investor Relations Service Providers</u> and such <u>optionsOptions</u> must vest in stages over <u>a period of not less than 12 months with such that:</u>
 - no more than 25% 1/4 of the Options vesting in anyvest no sooner than three month period months after the date the Options were granted;
 - (ii) no more than another 1/4 of the Options vest no sooner than six months after the date the Options were granted;
 - (iii) no more than another 1/4 of the Options vest no sooner than nine months after the date the Options were granted; and
 - (iv) the remainder of the Options vest no sooner than 12 months after the date the Options were granted,

and such limitation will not be an amendment to this Plan requiring the Option Holders consent under section 9.2 of this Plan.

2.4 Notification of Grant

Following the granting of an Option, the Administrator shall, within a reasonable period of time, notify the Option Holder in writing of the grant and shall enclose with such notice the Option Certificate representing the Option so granted. In no case will the Company be required to deliver an Option Certificate to an Option Holder until such time as the Company has obtained all necessary Regulatory Approvals for the grant of the Option.

2.5 Copy of Plan

Each Option Holder, concurrently with the notice of the grant of the Option, shall be provided with a copy of the Plan. A copy of any amendment to the Plan shall be promptly provided by the Administrator to each Option Holder.

2.6 Limitation on Service

The Plan does not give any Option Holder that is an Executive the right to serve or continue to serve as an Executive of the Company or any Subsidiary, nor does it give any Option Holder that is an Employee or Consultant the right to be or to continue to be employed or engaged by the Company or any Subsidiary.

2.7 No Obligation to Exercise

Option Holders shall be under no obligation to exercise Options granted under this Plan.

2.8 Agreement

The Company and every Option Holder granted an Option hereunder shall be bound by and subject to the terms and conditions of this Plan. By accepting an Option granted hereunder, the Option Holder has expressly agreed with the Company to be bound by the terms and conditions of this Plan. In the event that the Option Holder receives his, her or its Options pursuant to an oral or written agreement with the Company or a Subsidiary, whether such agreement is an employment agreement, consulting agreement or any other kind of agreement of any kind whatsoever, the Option Holder acknowledges that in the event of any inconsistency between the terms relating to the grant of such Options in that agreement and the terms attaching to the Options as provided for in this Plan, the terms provided for in this Plan shall prevail and the other agreement shall be deemed to have been amended accordingly.

2.9 Notice

Any notice, delivery or other correspondence of any kind whatsoever to be provided by the Company to an Option Holder will be deemed to have been provided if provided to the last home address, fax number or email address of the Option Holder in the records of the Company and the Company shall be under no obligation to confirm receipt or delivery.

2.10 Representation to TSXV

As a condition precedent to the issuance of an Option, the Company and the Option Holder must be able to represent to the TSXV as of the Grant Date that the Option Holder is a *bona fide* Executive, Employee or Consultant of the Company or any Subsidiary. The Option Certificate to which the Option Holder is a party must contain such a representation by the Option Holder.

SECTION 3 NUMBER OF SHARES UNDER PLAN

3.1 Board to Approve Issuance of Shares

The Board shall approve by resolution the issuance of all Shares to be issued to Option Holders upon the exercise of Options, such authorization to be deemed effective as of the Grant Date of such Options regardless of when it is actually done. The Board shall be entitled to approve the issuance of Shares in advance of the Grant Date, retroactively after the Grant Date, or by a general approval of this Plan.

3.2 Number of Shares

Subject to adjustment as provided for herein, the number of Shares which will be available for purchase pursuant to Options granted pursuant to this Plan, together with the number of Shares issuable at any time under all other security based compensation arrangements, will not exceed 10% of the number of Shares which are issued and outstanding on the particular date of grant of Options. If any Option expires or otherwise terminates for any reason without having been exercised in full, the number of Shares in respect of such expired or terminated Option shall again be available for the purposes of granting Options pursuant to this Plan.

3.3 Fractional Shares

No fractional shares shall be issued upon the exercise of any Option and, if as a result of any adjustment, an Option Holder would become entitled to a fractional share, such Option Holder shall have the right to purchase only the next lowest whole number of Shares and no payment or other adjustment will be made for the fractional interest.

SECTION 4 GRANT OF OPTIONS

4.1 Grant of Options

The Committee shall, from time to time in its sole discretion, grant Options to such Persons or Entities and on such terms and conditions as are permitted under this Plan.

4.2 Record of Option Grants

The Committee shall be responsible to maintain a record of all Options granted under this Plan and such record shall contain, in respect of each Option:

- (a) the name and address of the Option Holder;
- (b) the category (Executive, Employee or Consultant) under which the Option was granted to him, her or it;
- (c) the Grant Date and Expiry Date of the Option;
- (d) the number of Shares which may be acquired on the exercise of the Option and the Exercise Price of the Option;
- (e) the vesting and other additional terms, if any, attached to the Option; and
- (f) the particulars of each and every time the Option is exercised.

4.3 Effect of Plan

All Options granted pursuant to the Plan shall be subject to the terms and conditions of the Plan notwithstanding the fact that the Option Certificates issued in respect thereof do not expressly contain such terms and conditions but instead incorporate them by reference to the Plan. The Option Certificates will be issued for convenience only and in the case of a dispute with regard to any matter in respect thereof, the provisions of the Plan and the records of the Company shall prevail over the terms and conditions in the Option Certificate, save and except as noted below. Each Option will also be subject to, in addition to the provisions of the Plan, the terms and conditions contained in the schedules, if any, attached to the Option Certificate for such Option. Should the terms and conditions contained in such schedules be inconsistent with the provisions of the Plan, such terms and conditions will supersede the provisions of the Plan.

SECTION 5 TERMS AND CONDITIONS OF OPTIONS

5.1 Exercise Period of Option

Subject to sections 5.2, 5.5, 6.2, 6.3, 6.4 and 11.4, the Grant Date and the Expiry Date of an Option shall be the dates fixed by the Committee at the time the Option is granted and shall be set out in the Option Certificate issued in respect of such Option.

5.2 Extension of Expiry Date of Stock Options Due to a Black-Out

The Expiry Date of outstanding Options held by Options Option Holders which would expire during a Black-Out, or within 10 business days after the expiry of a Black-Out, will be extended for a period of time ending on the tenth (10th) business day after the expiry date of the Black-Out to provide such Options Holders with an extension to the right to exercise such Options; provided, however, that for so long as the Company is listed on the TSXV.

- (a) the Expiry Date must not exceed the date which is ten years from the date of grant of such Option;
- (b) the automatic extension of an Option Holder's Option pursuant to this section 5.2 will not be permitted where the Option Holder or the Company is subject to a cease trade order (or similar order under Securities Laws) in respect of the Company's securities; and
- (c) the automatic extension is available to all eligible Option Holders under this Plan under the same terms and conditions.

5.3 Number of Shares Under Option

The number of Shares which may be purchased pursuant to an Option shall be determined by the Committee and shall be set out in the Option Certificate issued in respect of the Option.

5.4 Exercise Price of Option

The Exercise Price at which an Option Holder may purchase a Share upon the exercise of an Option shall be determined by the Committee and shall be set out in the Option Certificate issued in respect of the Option. The Exercise Price shall not be less than the Market Value of the Shares as of the Grant Date. The Market Value of the Shares for a particular Grant Date shall be determined as follows:

- (a) for each organized trading facility on which the Shares are listed, Market Value will be the closing trading price of the Shares on the day immediately preceding the Grant Date, and may be less than this price if it is within the discounts permitted by the applicable Regulatory Authorities;
- (b) if the Company's Shares are listed on more than one organized trading facility, the Market Value shall be the Market Value as determined in accordance with subparagraph (a) above for the primary organized trading facility on which the Shares are listed, as determined by the Committee, subject to any adjustments as may be required to secure all necessary Regulatory Approvals;
- (c) if the Company's Shares are listed on one or more organized trading facilities but have not traded during the ten trading days immediately preceding the Grant Date, then the Market Value will be, subject to any adjustments as may be required to secure all necessary Regulatory Approvals, such value as is determined by the Committee; and
- (d) if the Company's Shares are not listed on any organized trading facility, then the Market Value will be, subject to any adjustments as may be required to secure all necessary Regulatory Approvals, such value as is determined by the Committee to be the fair value of the Shares, taking into consideration all factors that the Committee deems appropriate, including, without limitation, recent sale and offer prices of the Shares in private transactions negotiated at arms' length.

Notwithstanding anything else contained herein, in no case will the Market Value be less than the minimum prescribed by each of the organized trading facilities that would apply to the Company on the Grant Date in question.

5.5 <u>Termination of Option</u>

Subject to such other terms or conditions that may be attached to Options granted hereunder, an Option Holder may exercise an Option in whole or in part at any time and from time to time during the Exercise Period. Any Option or part thereof not exercised within the Exercise Period shall terminate and become null, void and of no effect as of the Expiry Time on the Expiry Date. The Expiry Date of an Option shall be the earlier of the date so fixed by the Committee at the time the Option is granted as set out in the Option Certificate and the date established, if applicable, in paragraphs (a) or (b) below or sections 6.2, 6.3, 6.4, or 11.4 of this Plan:

- (a) Ceasing to Hold Office In the event that the Option Holder holds his or her Option as an Executive and such Option Holder ceases to hold such position other than by reason of death or Disability, the Expiry Date of the Option shall be, unless otherwise determined by the Committee and expressly provided for in the Option Certificate, the 90th day following the date the Option Holder ceases to hold such position unless the Option Holder ceases to hold such position as a result of:
 - (i) ceasing to meet the qualifications set forth in the corporate legislation applicable to the Company;
 - (ii) a special resolution having been passed by the shareholders of the Company removing the Option Holder as a director of the Company or any Subsidiary; or
 - (iii) an order made by any Regulatory Authority having jurisdiction to so order;

in which case the Expiry Date shall be the date the Option Holder ceases to hold such position; OR

- (b) Ceasing to be Employed or Engaged In the event that the Option Holder holds his or her Option as an Employee or Consultant and such Option Holder ceases to hold such position other than by reason of death or Disability, the Expiry Date of the Option shall be, unless otherwise determined by the Committee and expressly provided for in the Option Certificate, the 90th day following the date the Option Holder ceases to hold such position, unless the Option Holder ceases to hold such position as a result of:
 - (i) termination for cause;
 - (ii) resigning his or her position; or
 - (iii) an order made by any Regulatory Authority having jurisdiction to so order;

in which case the Expiry Date shall be the date the Option Holder ceases to hold such position.

In the event that the Option Holder ceases to hold the position of Executive, Employee or Consultant for which the Option was originally granted, but comes to hold a different position as an Executive, Employee or Consultant prior to the expiry of the Option, the Committee may, in its sole discretion, choose to permit the Option to stay in place for that Option Holder with such Option then to be treated as being held by that Option Holder in his or her new position and such will not be considered to be an amendment to the Option in question requiring the consent of the Option Holder under section 9.2 of this Plan. Notwithstanding anything else contained herein, in no case will an Option be exercisable later than the Expiry Date of the Option.

5.6 <u>Vesting of Option and Acceleration</u>

The vesting schedule for an Option, if any, shall be determined by the Committee and shall be set out in the Option Certificate issued in respect of the Option. The Committee may elect, at any time, to accelerate the vesting schedule of one or more Options including, without limitation, on a Triggering Event, and such acceleration will not be considered an amendment to the Option in question requiring the consent of the Option Holder under section 9.2 of this Plan. For greater certainty, there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the TSXV.

5.7 <u>Additional Terms</u>

Subject to all applicable Regulatory Rules and all necessary Regulatory Approvals, the Committee may attach additional terms and conditions to the grant of a particular Option, such terms and conditions to be set out in a schedule attached to the Option Certificate. The Option Certificates will be issued for convenience only, and in the case of a dispute with regard to any matter in respect thereof, the provisions of this Plan and the records of the Company shall prevail over the terms and conditions in the Option Certificate, save and except as noted below. Each Option will also be subject to, in addition to the provisions of the Plan, the terms and conditions contained in the schedules, if any,

attached to the Option Certificate for such Option. Should the terms and conditions contained in such schedules be inconsistent with the provisions of the Plan, such terms and conditions will supersede the provisions of the Plan.

SECTION 6 TRANSFERABILITY OF OPTIONS

6.1 Non-transferable

Except as provided otherwise in this Section 6, Options are non-assignable and non-transferable.

6.2 Death of Option Holder

In the event of the Option Holder's death, any Options held by such Option Holder shall pass to the Personal Representative of the Option Holder and shall be exercisable by the Personal Representative on or before the date which is the earlier of one year following the date of death and the applicable Expiry Date.

6.3 Disability of Option Holder

If the employment or engagement of an Option Holder as an Employee or Consultant or the position of an Option Holder as a director Director or officer Officer of the Company or a Subsidiary is terminated by the Company by reason of such Option Holder's Disability, any Options held by such Option Holder shall be exercisable by such Option Holder or by the Personal Representative on or before the date which is the earlier of one year following the termination of employment, engagement or appointment as a director Director or officer Officer and the applicable Expiry Date.

6.4 <u>Disability and Death of Option Holder</u>

If an Option Holder has ceased to be employed, engaged or appointed as a director Director or officer Officer of the Company or a Subsidiary by reason of such Option Holder's Disability and such Option Holder dies within one year after the termination of such engagement, any Options held by such Option Holder that could have been exercised immediately prior to his or her death shall pass to the Personal Representative of such Option Holder and shall be exercisable by the Personal Representative on or before the date which is the earlier of one year following the death of such Option Holder and the applicable Expiry Date.

6.5 Vesting

Notwithstanding any vesting schedule to which Options are subject, Options shall cease to vest immediately if the employment or engagement of an Option Holder as an Employee or Consultant or the position of an Option Holder as a director Director or officer Officer of the Company or a Subsidiary is terminated for any reason whatsoever. In which case, the Option Holder may only exercise such number of Options that are vested as at the date of termination of such Option Holder's employment, engagement or appointment as a director Director or officer Officer.

6.6 Deemed Non-Interruption of Engagement

Employment or engagement by the Company shall be deemed to continue intact during any military or sick leave or other *bona fide* leave of absence if the period of such leave does not exceed 90 days or, if longer, for so long as the Option Holder's right to re-employment or re-engagement by the Company is guaranteed either by statute or by contract. If the period of such leave exceeds 90 days and the Option Holder's re-employment or re-engagement is not so guaranteed, then his or her employment or engagement shall be deemed to have terminated on the ninety-first day of such leave.

SECTION 7 EXERCISE OF OPTION

7.1 Exercise of Option

An Option may be exercised only by the Option Holder or the Personal Representative of any Option Holder. An Option Holder or the Personal Representative of any Option Holder may exercise an Option in whole or in part at any time and from time to time during the Exercise Period up to the Expiry Time on the Expiry Date by delivering to the Administrator the required Exercise Notice, the applicable Option Certificate and a certified cheque, wire transfer or bank draft payable to the Company in an amount equal to the aggregate Exercise Price of the Shares then being purchased pursuant to the exercise of the Option. Notwithstanding anything else contained herein, Options may not be exercised during Black-Out unless the Committee determines otherwise. Notwithstanding anything else contained herein, Options held by Investor Relations Service Providers may not be exercised on a "net exercise" basis.

7.2 Issue of Share Certificates

As soon as reasonably practicable following the receipt of the Exercise Notice, the Administrator shall cause to be delivered to the Option Holder a certificate for the Shares so purchased. If the number of Shares so purchased is less than the number of Shares subject to the Option Certificate surrendered, the Administrator shall also provide a new Option Certificate for the balance of Shares available under the Option to the Option Holder concurrent with delivery of the Share Certificate.

7.3 No Rights as Shareholder

Until the date of the issuance of the certificate for the Shares purchased pursuant to the exercise of an Option, no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to such Shares, notwithstanding the exercise of the Option, unless the Committee determines otherwise. In the event of any dispute over the date of the issuance of the certificates, the decision of the Committee shall be final, conclusive and binding.

SECTION 8 ADMINISTRATION

8.1 Board or Committee

The Plan shall be administered by the Board, by a Committee of the Board appointed in accordance with section 8.2 below, or by an Administrator appointed in accordance with subsection 8.4(b).

8.2 Appointment of Committee

The Board may at any time appoint a Committee, consisting of not less than two of its members, to administer the Plan on behalf of the Board in accordance with such terms and conditions as the Board may prescribe, consistent with this Plan. Once appointed, the Committee shall continue to serve until otherwise directed by the Board. From time to time, the Board may increase the size of the Committee and appoint additional members, remove members (with or without cause) and appoint new members in their place, fill vacancies however caused, or remove all members of the Committee and thereafter directly administer the Plan.

8.3 Quorum and Voting

A majority of the members of the Committee shall constitute a quorum and, subject to the limitations in this Section 8, all actions of the Committee shall require the affirmative vote of members who constitute a majority of such quorum. Members of the Committee may vote on any matters affecting the administration of the Plan or the grant of Options pursuant to the Plan, except that no such member shall act upon the granting of an Option to himself or herself (but any such member may be counted in determining the existence of a quorum at any meeting of the Committee during which

action is taken with respect to the granting of Options to that member). The Committee may approve matters by written resolution signed by a majority of the quorum.

8.4 Powers of Committee

The Committee (or the Board if no Committee is in place) shall have the authority to do the following:

- (a) administer the Plan in accordance with its terms;
- (b) appoint or replace the Administrator from time to time;
- (c) determine all questions arising in connection with the administration, interpretation and application of the Plan, including all questions relating to the Market Value of the Shares;
- (d) correct any defect, supply any information or reconcile any inconsistency in the Plan in such manner and to such extent as shall be deemed necessary or advisable to carry out the purposes of the Plan;
- (e) prescribe, amend, and rescind rules and regulations relating to the administration of the Plan;
- (f) determine the duration and purposes of leaves of absence from employment or engagement by the Company which may be granted to Option Holders without constituting a termination of employment or engagement for purposes of the Plan;
- (g) do the following with respect to the granting of Options:
 - (i) determine the Executives, Employees or Consultants to whom Options shall be granted, based on the eligibility criteria set out in this Plan;
 - (ii) determine the terms of the Option to be granted to an Option Holder including, without limitation, the Grant Date, Expiry Date, Exercise Price and vesting schedule (which need not be identical with the terms of any other Option);
 - (iii) subject to any necessary Regulatory Approvals and section 9.2, amend the terms of any Options;
 - (iv) determine when Options shall be granted; and
 - (v) determine the number of Shares subject to each Option;
- (h) accelerate the vesting schedule of any Option previously granted, provided that there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the TSXV; and
- (i) make all other determinations necessary or advisable, in its sole discretion, for the administration of the Plan.

8.5 Administration by Committee

All determinations made by the Committee in good faith shall be final, conclusive and binding upon all persons. The Committee shall have all powers necessary or appropriate to accomplish its duties under this Plan.

8.6 Interpretation

The interpretation by the Committee of any of the provisions of the Plan and any determination by it pursuant thereto shall be final, conclusive and binding and shall not be subject to dispute by any Option Holder. No member of the

Committee or any person acting pursuant to authority delegated by it hereunder shall be personally liable for any action or determination in connection with the Plan made or taken in good faith and each member of the Committee and each such person shall be entitled to indemnification with respect to any such action or determination in the manner provided for by the Company.

SECTION 9 APPROVALS AND AMENDMENT

9.1 Shareholder Approval of Plan

If required by a Regulatory Authority or by the Committee, this Plan may be made subject to the approval of a majority of the votes cast at a meeting of the shareholders of the Company or by a majority of votes cast by disinterested shareholders at a meeting of shareholders of the Company. If shareholder approval is required, any Options granted under this Plan prior to such time will not be exercisable or binding on the Company unless and until such shareholder approval is obtained.

9.2 Amendment of Option or Plan

Subject to any required Regulatory Approvals, the Committee may from time to time amend any existing Option or the Plan or the terms and conditions of any Option thereafter to be granted provided that where such amendment relates to an existing Option and it would:

- (a) materially decrease the rights or benefits accruing to an Option Holder; or
- (b) materially increase the obligations of an Option Holder;

then, unless otherwise excepted out by a provision of this Plan, the Committee must also obtain the written consent of the Option Holder in question to such amendment. If at the time the Exercise Price Disinterested Shareholder Approval is required for any reduction in the exercise price of an Option is reduced, or the extension of the term of an Option, if the Option Holder is an Insider of the Company, the Insider must not exercise the option at the reduced Exercise Price until the reduction in Exercise Price has been approved by the disinterested shareholders of the Company, if required by the TSXV at the time of the proposed amendment.

SECTION 10 CONDITIONS PRECEDENT TO ISSUANCE OF OPTIONS AND SHARES

10.1 Compliance with Laws

An Option shall not be granted or exercised, and Shares shall not be issued pursuant to the exercise of any Option, unless the grant and exercise of such Option and the issuance and delivery of such Shares comply with all applicable Regulatory Rules, and such Options and Shares will be subject to all applicable trading restrictions in effect pursuant to such Regulatory Rules and the Company shall be entitled to legend the Option Certificates and the certificates representing such Shares accordingly.

10.2 Obligation to Obtain Regulatory Approvals

In administering this Plan, the Committee will seek any Regulatory Approvals which may be required. The Committee will not permit any Options to be granted without first obtaining the necessary Regulatory Approvals unless such Options are granted conditional upon such Regulatory Approvals being obtained. The Committee will make all filings required with the Regulatory Authorities in respect of the Plan and each grant of Options hereunder. No Option granted will be exercisable or binding on the Company unless and until all necessary Regulatory Approvals have been obtained. The Committee shall be entitled to amend this Plan and the Options granted hereunder in order to secure any necessary Regulatory Approvals and such amendments will not require the consent of the Option Holders under section 9.2 of this Plan.

10.3 Inability to Obtain Regulatory Approvals

The Company! is inability to obtain Regulatory Approval from any applicable Regulatory Authority, which Regulatory Approval is deemed by the Committee to be necessary to complete the grant of Options hereunder, the exercise of those Options or the lawful issuance and sale of any Shares pursuant to such Options, shall relieve the Company of any liability with respect to the failure to complete such transaction.

10.4 Withholding Tax Requirements

Upon exercise of an Option, the Option Holder shall, upon notification of the amount due and prior to the delivery of the certificates representing the Shares, pay to the Company amounts necessary to satisfy applicable federal and provincial withholding tax requirements and, if applicable, Canada Pension Plan contributions, in such amount as determined by the Company, or shall otherwise make arrangements satisfactory to the Company for such requirements. In order to implement this provision, the Company or any related corporation shall have the right to retain and withhold from any payment of cash or Shares under this Plan the amount of taxes and, if applicable, Canada Pension Plan contributions, in such amount as determined by the Company, to be withheld or otherwise deducted and paid with respect to such payment. At its discretion, the Company may require an Option Holder receiving Shares to reimburse the Company for any such taxes and Canada Pension Plan contributions required to be withheld by the Company and withhold any distribution to the Option Holder in whole or in part until the Company is so reimbursed. In lieu thereof, the Company shall have the right to withhold from any other cash amounts due or to become due from the Company to the Option Holder an amount equal to such taxes and, if applicable, Canada Pension Plan contributions as determined by the Company. The Company may also retain and withhold or the Option Holder may elect, subject to approval by the Company at its sole discretion, to have the Company retain and withhold a number of Shares having a market value of not less than the amount of such taxes and, if applicable, Canada Pension Plan contributions, as determined by the Company, required to be withheld by the Company to reimburse the Company for any such taxes and cancel (in whole or in part) any such Shares so withheld.

SECTION 11 ADJUSTMENTS AND TERMINATION

11.1 <u>Termination of Plan</u>

Subject to any necessary Regulatory Approvals, the Committee may terminate or suspend the Plan. Unless earlier terminated as provided in this Section 11, the Plan shall terminate on, and no more Options shall be granted under the Plan after, the tenth anniversary of the Effective Date of the Plan.

11.2 No Grant During Suspension of Plan

No Option may be granted during any suspension, or after termination, of the Plan. Suspension or termination of the Plan shall not, without the consent of the Option Holder, alter or impair any rights or obligations under any Option previously granted.

11.3 Alteration in Capital Structure

- (a) Following the date an Option is granted, the exercise price for and the number of Shares which are subject to an Option will be adjusted, with respect to the then unexercised portion thereof, in the events and in accordance with the provisions and rules set out in this section 11.3, with the intent that the rights of Option Holders under their Options are, to the extent possible, preserved and maintained notwithstanding the occurrence of such events. Any dispute that arises at any time with respect to any adjustment pursuant to such provisions and rules will be conclusively determined by the Board or Committee, and any such determination will be binding on the Company, the Option Holder and all other affected parties.
- (b) If the outstanding Shares are changed into or exchanged for a different number of shares or into or for other securities of the Company or securities of another company or entity, whether through an arrangement,

amalgamation, merger, business combination, sale or other similar procedure or otherwise, or a share recapitalization, subdivision or consolidation, then on each exercise of the Option which occurs following such events, for each Share for which the Option is exercised, the Option Holder shall instead receive the number and kind of shares or other securities of the Company or other company into which such Share would have been changed or for which such Share would have been exchanged if it had been outstanding on the date of such event and the exercise price will be similarly adjusted so that the aggregate price to exercise the Option is preserved, and if the Company undertakes an arrangement or is amalgamated, merged or combined with another company, the Board shall make such other provision for the protection of the rights of Option Holders as it shall deem advisable.

- (c) If the outstanding Shares are changed into or exchanged for a different number of shares or into or for other securities of the Company or securities of another company or entity, in a manner other than as specified in subsection 11.3(b) then the Board or Committee, in its sole discretion, may make such adjustment to the securities to be issued pursuant to any exercise of the Option and the exercise price to be paid for each such security following such event as the Board or Committee in its sole and absolute discretion determines to be equitable to give effect to the principle described in subsection 11.3(a) and such adjustments shall be effective and binding upon the Company and the Option Holder and all the other parties for all purposes.
- (d) No adjustment or substitution provided for in this section 11.3 shall require the Company to issue a fractional share in respect of any Option. Fractional shares shall be eliminated.
- (e) The grant or existence of an Option shall not in any way limit or restrict the right or power of the Company to effect adjustments, reclassifications, reorganizations, arrangements or changes of its capital or business structure, or to amalgamate, merge, consolidate, dissolve or liquidate, or to sell or transfer all or any part of its business or assets.
- (f) For purposes of this section 11.3, and without limitation, neither:
 - (i) the issuance of additional securities of the Company in exchange for adequate consideration (including services); nor
 - (ii) the conversion of outstanding securities of the Company into Shares,
 - shall trigger any adjustment pursuant to this section 11.3.
- (g) Any adjustment made to any Options pursuant to this section 11.3 shall not be considered an amendment requiring the Option Holder's consent for the purposes of Section 9.2 of this Plan.
- (h) For greater certainty, any adjustment, other than in connection with a share consolidation or share split, to
 Options granted or issued under this Plan are subject to the prior acceptance of the TSXV, including
 adjustments related to an amalgamation, merger, arrangement, reorganization, spin-off, dividend or
 recapitalization.

11.4 Triggering Events

Subject to the Company complying with section 11.5 and any necessary Regulatory Approvals and notwithstanding any other provisions of this Plan or any Option Certificate, the Committee may, without the consent of the Option Holder or Holders in question:

(a) cause all or a portion of any of the Options granted under the Plan to terminate upon the occurrence of a Triggering Event; or

(b) cause all or a portion of any of the Options granted under the Plan to be exchanged for incentive stock options of another corporation upon the occurrence of a Triggering Event in such ratio and at such exercise price as the Committee deems appropriate, acting reasonably.

Such termination or exchange shall not be considered an amendment requiring the Option Holder's consent for the purpose of section 9.2 of the Plan.

11.5 Notice of Termination by Triggering Event

In the event that the Committee wishes to cause all or a portion of any of the Options granted under this Plan to terminate on the occurrence of a Triggering Event, it must give written notice to the Option Holders in question not less than 10 days prior to the consummation of a Triggering Event so as to permit the Option Holder the opportunity to exercise the vested portion of the Options prior to such termination. Upon the giving of such notice and subject to any necessary Regulatory Approvals, all Options or portions thereof granted under the Plan which the Company proposes to terminate shall become immediately exercisable notwithstanding any contingent vesting provision to which such Options may have otherwise been subject. For greater certainty, there shall be no acceleration of the vesting requirements applicable to Options granted to Investor Relations Service Providers without the prior written approval of the TSXV.

11.6 Determinations to be Made By Committee

Adjustments and determinations under this Section 11 shall be made by the Committee, whose decisions as to what adjustments or determination shall be made, and the extent thereof, shall be final, binding, and conclusive.

11.7 Options Granted to U.S. Residents or Citizens

Any Option granted under the Plan to an Option Holder who is a citizen or resident of the United States (including its territories, possessions and all areas subject to the jurisdiction) (a "": U.S. Option Holder") may be an incentive stock option (an "": SO") within the meaning of Section 422 of the Internal Revenue Code of 1986, as amended, of the United States (the ""Code"), but only if so designated by the Company in the agreement evidencing such Option. No provision of this Plan, as it may be applied to a U.S. Option Holder with respect to Options which are designated as ISO! s, shall be construed so as to be inconsistent with any provision of Section 422 of the Code. Grants of Options to U.S. Option Holders pursuant to Section 2 hereof which are not designated as or otherwise do not qualify as ISO! s will be treated as non-statutory stock options for U.S. federal tax purposes. Notwithstanding anything in this Plan contained to the contrary, the following provisions shall apply to ISO! s granted to each U.S. Option Holder:

- (a) ISO's shall only be granted to individual U.S. Option Holders who are, at the time of grant, employees of the Company (within the meaning of the Code). Any director of the Company who is a U.S. Option Holder shall be ineligible to vote upon the granting of such Option;
- (b) the aggregate fair market value (determined as of the time an ISO is granted) of the Shares subject to ISO's exercisable for the first time by a U.S. Option Holder during any calendar year under this Plan and all other stock option plans, within the meaning of Section 422 of the Code, of the Company shall not exceed One Hundred Thousand Dollars in U.S. funds (U.S.\$100,000);
- (c) the Exercise Price for Shares under each ISO granted to a U.S. Option Holder pursuant to this Plan shall be not less than the fair market value of such Shares at the time granted, as determined in good faith by the directors at such time (unless such ISO is granted pursuant to an assumption or substitution for another option in a manner satisfying the provisions of Section 424(a) of the Code);
- (d) if any U.S. Option Holder to whom an ISO is to be granted under the Plan at the time of the grant of such ISO is the owner of shares possessing more than ten percent (10%) of the total combined voting power of all classes of shares of the Company, then the following special provisions shall be applicable to the ISO granted to such individual:

- (a) (i) the Exercise Price (per Share) subject to such ISO shall not be less than one hundred ten percent (110%) of the fair market value of one Share at the time of grant; and
- (ii) for the purposes of this Section 11.7(d)(ii) only, the exercise period shall not exceed five (5) years from the date of grant;
- (e) no ISO may be granted hereunder to a U.S. Option Holder following the expiration of ten (10) years after the date on which this Plan is adopted by the Company or the date on which the Plan is approved by the shareholders of the Company, whichever is earlier;
- (f) no ISO granted to a U.S. Option Holder under the Plan shall become exercisable unless and until the Plan shall have been approved by the shareholders of the Company;
- (g) no ISO shall be transferable by a U.S. Option Holder other than by will or the laws of descent and distribution; and
- (h) during the lifetime of the original grantee of an ISO, such ISO may not be exercised by anyone other than such grantee.

SCHEDULE "A"

[Include the following Exchange hold period for stock options granted to: (i) directors, officers and promoters; (ii) over 10% shareholders; and (iii) any Option Holder if the exercise price of the stock options granted is based on less than Market Price.]

[Without prior written approval of the TSX Venture Exchange and compliance with all applicable securities legislation, the securities represented by this certificate may not be sold, transferred, hypothecated or otherwise traded on or through the facilities of the TSX Venture Exchange or otherwise in Canada or to or for the benefit of a Canadian resident until ●[date four months and one day after Grant Date].]

SALAZAR RESOURCES LIMITED

STOCK OPTION PLAN — OPTION CERTIFICATE

This Option Certificate is issued pursuant to the provisions of the Stock Option Plan (the "Plan") of Salazar Resources Limited (the "Company") and evidences that ●[Name of Option Holder] is the holder (the "Option Holder") of an option (the "Option") to purchase up to ● common shares (the "Shares") in the capital stock of the Company at a purchase price of Cdn.\$● per Share (the "Exercise Price"). This Option may be exercised at any time and from time to time from and including the following Grant Date through to and including up to 5:00 p.m. local time in Vancouver, British Columbia (the "Expiry Time") on the following Expiry Date:

- (a) the Grant Date of this Option is \bullet , 20 \bullet ; and
- (b) subject to sections 5.2, 5.5, 6.2, 6.3, 6.4 and 11.4 of the Plan, the Expiry Date of this Option is \bullet , $20\bullet$

To exercise this Option, the Option Holder must deliver to the Administrator of the Plan, prior to the Expiry Time on the Expiry Date, an Exercise Notice, in the form provided in the Plan, which is incorporated by reference herein, together with the original of this Option Certificate and a certified cheque or bank draft payable to the Company in an amount equal to the aggregate of the Exercise Price of the Shares in respect of which this Option is being exercised.

This Option Certificate and the Option evidenced hereby is not assignable, transferable or negotiable and is subject to the detailed terms and conditions contained in the Plan. This Option Certificate is issued for convenience only and in the case of any dispute with regard to any matter in respect hereof, the provisions of the Plan and the records of the Company shall prevail. This Option is also subject to the terms and conditions contained in the schedules, if any, attached hereto.

[Include the following Exchange hold period for stock options granted to: (i) directors, officers and promoters; (ii) over 10% shareholders; and (iii) any Option Holder if the exercise price of the stock options granted is based on less than Market Price.]

[Any share certificates issued pursuant to an exercise of the Option before • [date four months and one day after Grant Date] will contain the following legend:

"Without prior written approval of the TSX Venture Exchange and compliance with all applicable securities legislation, the securities represented by this certificate may not be sold, transferred, hypothecated or otherwise traded on or through the facilities of the TSX Venture Exchange or otherwise in Canada or to or for the benefit of a Canadian resident until ●[date four months and one day after Grant Date]."]

If the Option Holder is a resident or citizen of the United States of America at the time of the exercise of the Option, the certificate(s) representing the Shares will be endorsed with the following or a similar legend:

"The securities represented by this certificate have not been registered under the Securities Act of 1933, as amended, of the United States of America (the "Act") or the securities laws of any state ("State") of the United States of America and may not be sold, transferred, pledged, hypothecated or distributed, directly or indirectly, to a U.S. person (as defined in Regulation S adopted by the U.S.

Securities and Exchange Commission under the Act) or within the United States unless such securities are (i) registered under the Act and any applicable State securities act (a "State Act"), or (ii) exempt from registration under the Act and any applicable State Act and the Company has received an opinion of counsel to such effect reasonably satisfactory to it, or (iii) sold in accordance with Regulation S and the Company has received an opinion of counsel to such effect reasonably satisfactory to it."

This Option was granted to the Option Holder in his or her capacity as a bona fide Director, Officer, <u>Employee</u>, <u>Management Company</u> Employee or Consultant of the Company, and shall continue in effect should his or her status change and he or she continue in a new capacity as a Director, Officer, <u>Employee</u>, <u>Management Company</u> Employee or Consultant of the Company.

SALAZAR RESOURCES LIMITED

SALAZAK K	ESOURCES LIMITED				
Per:					
Direc	etor				
is a bona fide appropriate r accepts this Op file and otherw Option and ex further acknow	older acknowledges receipt of a copy of the Plan and represents to the Company that the Option Holder Director, Officer, Employee, Management Company Employee or Consultant of the Company (circle relationship with the Company) and is familiar with the terms and conditions of the Plan, and hereby ption subject to all of the terms and conditions of the Plan. The Option Holder agrees to execute, deliver, wise assist the Company in filing any report, undertaking or document with respect to the awarding of the ercise of the Option, as may be required by the applicable Regulatory Authorities. The Option Holder yieldges that if the Plan has not been approved by the shareholders of the Company on the Grant Date, this exercisable until such approval has been obtained.				
By signing this	s Option Certificate, the undersigned also provides its express written consent to:				
(a)	the disclosure of Personal Information (as defined below) by the Company to the TSX Venture Exchange (the "Exchange") with respect to any and all forms required to be filed by the Company with the Exchange with respect to the grant of this Option; and				
(b)	the collection, use and disclosure of Personal Information by the Exchange for the purposes described in Appendix 6A of the Corporate Finance Manual of the Exchange, or as otherwise identified by the Exchange, from time to time.				
	rmation" means any information about an identifiable individual, and includes the information contained G – Summary Form – Incentive Stock Options to be filed by the Company with the Exchange.				
Signature of C	•				
Signature	Date signed:				
Print Name					
Address					

OPTION CERTIFICATE – SCHEDULE

[Complete the following additional terms and any other special terms, if applicable, or remove the inapplicable terms or this schedule entirely.]

The additional terms and conditions attached to the Option represented by this Option Certificate are as follows:

- 1. The Options will not be exercisable unless and until they have vested and then only to the extent that they have vested. The Options will vest in accordance with the following:
 - (a) <u>no more than</u> Shares (●%) will vest and be exercisable on or after the Grant Date;
 - (b) no more than additional Shares (●%) will vest and be exercisable on or after [date];
 - (c) <u>no more than</u> additional Shares (●%) will vest and be exercisable on or after [date]; <u>and</u>
 - (d) <u>the remainder of the</u> additional Shares (•%) will vest and be exercisable on or after [date];
- 2. Upon the Option Holder ceasing to hold a position with the Company, other than as a result of the events set out in paragraphs 5.5(a) or 5.5(b) of the Plan, the Expiry Date of the Option shall be [Insert date desired that is longer or shorter than the standard 90 days as set out in the Plan] following the date the Option Holder ceases to hold such position.

SCHEDULE "B"

SALAZAR RESOURCES LIMITED **STOCK OPTION PLAN**

NOTICE OF EXERCISE OF OPTION

ТО:	The Administrator, Stock Opt Salazar Resources Limited Suite 1305, 1090 West Georg Vancouver, British Columbia (or such other address as the O	gia Street V6E 3V7			
	ned hereby irrevocably gives notic 'Company"), of the exercise of the				
(a) all of	f the Shares; or				
(b)	of the Shares;				
which are the	subject of the Option Certificate a	attached hereto (attach your o	original Opti	ion Certificate)	١.
in an amount certificate evi	ned tenders herewith a certified che equal to the aggregate Exercise idencing said Shares in the name ride full complete address):	Price of the aforesaid Shares	and directs	the Company	to issue the
with this forr Vancouver, B	ned acknowledges the Option is no n and delivered to the required s.C. on the Expiry Date of the Opti	address with the required partion.			
		Signature	of	Option	Holder

TO:

Schedule "C"

SALARAR RESOURCES LTD.

RESTRICTED SHARE UNIT PLAN

ARTICLE I DEFINITIONS AND INTERPRETATION

1.1 Definitions

For purposes of this Plan:

- (a) "Account" means an account maintained by the Company for each Participant and which will be credited with RSUs in accordance with the terms of this Plan;
- (b) "Award Date" means the date or dates on which an award of RSUs is made to a Participant in accordance with Section 4.1;
- (c) "Award Value" means, with respect to any RSUs, an amount equal to the number of RSUs, as such number may be adjusted in accordance with the terms of this Plan, multiplied by the Fair Market Value of the Shares;
- (d) "Black-Out Period" means the period of time when, pursuant to any policies of the Company, any securities of the Company may not be traded by certain persons as designated by the Company, including any Participant that holds an RSU;
- (e) "Black-Out" means a restriction imposed by the Company pursuant to the Company's internal policies on all or any of its Directors, Officers, Employees, Insiders or persons in a special relationship as a result of the bona fide existence of undisclosed Material Information whereby they are to refrain from trading in the Company's securities until the restriction has been lifted by the Company. Such restriction must expire following the general disclosure of the undisclosed Material Information;
- (f) (e) "Board" means the board of directors of the Company as constituted from time to time:
- (g) (f)-"Change of Control" means:
 - (i) a successful takeover bid; or
 - (ii) (A) any change in the beneficial ownership or control of the outstanding securities or other interests of the Company which results in:
 - (1) a person or group of persons "acting jointly or in concert" (within the meaning of NI 62-104); or
 - (2) an affiliate or associate of such person or group of persons;
 - holding, owning or controlling, directly or indirectly, more than 50% of the outstanding voting securities or interests of the Company; and
 - (B) members of the Board who are members of the Board immediately prior to the earlier of such change and the first public announcement of such

change cease to constitute a majority of the Board at any time within sixty days of such change; or

- (iii) Incumbent Directors no longer constituting a majority of the Board; or
- (iv) the winding up of the Company or the sale, lease or transfer of all or substantially all of the assets to any other person or persons (other than pursuant to an internal reorganization or in circumstances where the business of the Company is continued and where the shareholdings or other securityholdings, as the case may be, in the continuing entity and the constitution of the board of directors or similar body of the continuing entity is such that the transaction would not be considered a "Change of Control" if paragraph 1.1(f)(ii)) above was applicable to the transaction); or
- (v) any determination by a majority of the Board that a Change of Control has occurred or is about to occur and any such determination shall be binding and conclusive for all purposes of this Plan;
- (h) (g) "Code" means the U.S. Internal Revenue Code of 1986, as amended;
- (i) (h) "Committee" has the meaning ascribed thereto in Section 2.4;
- (i) "Company" means Salazar Resources Ltd., and includes any successor corporation thereof;
- (k) "Consultant" means, in relation to the Company, an individual (other than a Director, Officer or Employee of the Company) who:
 - (i) is engaged to provide, on an ongoing bona fide basis, consulting, technical, management or other services to the Company or any Subsidiary other than services provided in relation to a "distribution" (as that term is described in the Securities Act);
 - (ii) provides the services under a written contract between the Company or any Subsidiary and the individual or a Consultant Entity (as defined in clause (h)(iv) below); and
 - (iii) in the reasonable opinion of the Company, spends or will spend a significant amount of time and attention on the affairs and business of the Company or any Subsidiary:

and includes:

- (iv) a corporation of which the individual is an employee or shareholder or a partnership of which the individual is an employee or partner (a "Consultant Entity"); or
- (v) an RRSP or RRIF established by or for the individual under which he or she is the beneficiary;
- (I) "Director" means a director (as defined under Securities Laws) of the Company or any of its Subsidiaries;

- (m) "Disinterested Shareholder Approval" has the meaning ascribed thereto by the TSXV in "Policy 4.4 Security Based Compensation Arrangements" of the TSXV's Corporate Finance Manual.
- (n) "Dividend Equivalent" has the meaning ascribed thereto in Section 4.2;
- (o) (k) "Dividend Market Value" means the Fair Market Value per Share on the dividend record date;

(p) "Employee" means:

- (i) an individual who works full-time or part-time for the Company or of its Subsidiary and such other individual as may, from time to time, be permitted by applicable Regulatory Rules to be granted Options as an employee or as an equivalent thereto; or
- (ii) an individual who works for the Company or its Subsidiary either full-time or on a continuing and regular basis for a minimum amount of time per week providing services normally provided by an employee and who is subject to the same control and direction by the Company or its Subsidiary over the details and methods of work as an employee of the Company or of the Subsidiary, as the case may be, but for whom income tax deductions are not made at source.

and includes:

- (iii) a corporation wholly-owned by such individual; and
- (iv) any RRSP or RRIF established by or for such individual under which he or she is the beneficiary;
- (g) (I)-"Exchange" means the TSXV or, if the Shares are not then listed and posted for trading on the TSXV, such stock exchange on which such Shares are listed and posted for trading as may be selected for such purpose by the Board;
- (r) (m) "Expiry Date" means, with respect to a RSU, the expiry date as may be determined by the Board, in its sole discretion, and set out in the applicable RSU Agreement;
- (n) "Fair Market Value" with respect to a Share, as at any date, means the volume weighted average of the prices at which the Shares traded on the TSXV (or, if the Shares are not then listed and posted for trading on the TSXV or are then listed and posted for trading on more than one stock exchange, on such stock exchange on which the majority of the trading volume and value of the Shares occurs) for the three (3) trading days on which the Shares traded on the said exchange immediately preceding such date. In the event that the Shares are not listed and posted for trading on any stock exchange, the Fair Market Value shall be the fair market value of the Shares as determined by the Board in its sole discretion, acting reasonably and in good faith;
- (t) (e)—"Forfeiture Date" means the date that is the earlier of: (i) the effective date of the Participant's termination or resignation, as the case may be; and (ii) the date that the Participant ceases to be in the active performance of the usual and customary day-to-day duties of the Participant's position or job, regardless of whether adequate or proper advance notice of termination or resignation shall have been provided in respect of such cessation of being a Participant;

- (u) (p) "Incumbent Directors" means any member of the Board who was a member of the Board at the effective date of this Plan and any successor to an Incumbent Director who was recommended or elected or appointed to succeed any Incumbent Director by the affirmative vote of the Board, including a majority of the Incumbent Directors then on the Board, prior to the occurrence of the transaction, transactions, elections or appointments giving rise to a Change of Control;
- (v) (q) "Insider", "associate" and "affiliate" each have the meaning ascribed thereto in the TSX Venture Exchange Corporate Finance Manual, as amended from time to time;
- (w) "Investor Relations Service Provider" has the meaning ascribed to such term under the policy 4.4 of the Exchange;
- (x) "Management Company Employee" means an individual employed by a company providing management services to the Company, which services are required for the ongoing successful operation of the business enterprise of the Company;
- (y) "Material Change" means the definition prescribed by applicable Securities Laws;
- (z) "Material Fact" means the definition prescribed by applicable Securities Laws:
- (aa) "Material Information" means a Material Fact and/or Material Change as defined by applicable Securities Laws and Exchange Policy:
- (bb) (r) "NI 62-104" means National Instrument 62-104 Take-Over Bids and Issuer Bids, as amended from time to time;
- (cc) "Officer" means an officer (as defined under Securities Laws) of the Company or of any of its Subsidiaries;
- (dd) (s) "Outside Payment Date", in respect of a RSU, means December 31 of the calendar year in which the Expiry Date occurs;
- (ee) (t) "Participant" means any director, officer, employee or consultant Director, Officer, Employee, Consultant or Management Company Employee (excluding an investor relations consultant Investor Relations Service Provider) of, or a person or company engaged by, one or more of the entities comprising the Salazar Group to provide services for an initial, renewable or extended period, determined to be eligible to participate in this Plan in accordance with Section 3.1 and, where applicable, a former Participant deemed eligible to continue to participate in this Plan in accordance with Section 4.5;
- (ff) (u) "Plan" means this Restricted Share Unit Plan;
- (gg) (v)—"**RSU**" means a unit equivalent in value to a Share credited by means of a bookkeeping entry in the Participants' Accounts;
- (hh) (w) "RSU Agreement" has the meaning set forth in Section 3.2;
- (ii) (x) "Salazar Group" means, collectively, the Company, any entity that is a Subsidiary of the Company from time to time, and any other entity designated by the Board from time to time as a member of the Salazar Group for the purposes of this Plan (and, for greater certainty, including any successor entity of any of the aforementioned entities);

- (jj) "Security Based Compensation Arrangements" means any incentive plan of the Company (other than this Plan), including the Company's stock option plan, and any incentive options granted by the Company outside of this Plan;
- (kk) "Securities Laws" means securities legislation, securities regulation and securities rules, as amended, and the policies, notices, instruments and blanket orders in force from time to time that are applicable to the Company;
- (II) (z) "Share" means a common share of the Company;
- (mm) (aa) "Subsidiary" has the meaning ascribed thereto in the Securities Act (British Columbia);
- (nn) (bb) "Successor" has the meaning ascribed thereto in Section 5.2;
- (oo) (cc) "takeover bid" means a "take-over bid" as defined in NI 62-104 pursuant to which the "offeror" would as a result of such takeover bid, if successful, beneficially own, directly or indirectly, in excess of 50% of the outstanding Shares;
- (pp) (dd) "TSXV" means the TSX Venture Exchange Inc.;
- (qq) (ee) "U.S. Participant" means a Participant who is a citizen or resident of the United States (including its territories, possessions and all areas subject to the jurisdiction); and
- (ff)—"Vesting Date" means, with respect to any RSU, the date upon which the Award Value to which the Participant is entitled pursuant to such RSU shall irrevocably vest and become irrevocably payable by the Company to the Participant in accordance with the terms hereof.

1.2 Interpretation

Words in the singular include the plural and words in the plural include the singular. Words importing male persons include female persons, corporations or other entities, as applicable. The headings in this document are for convenience and reference only and shall not be deemed to alter or affect any provision hereof. The words "hereto", "herein", "hereby", "hereunder", "hereof" and similar expressions mean or refer to this document as a whole and not to any particular Article, Section, paragraph or other part hereof.

ARTICLE II PURPOSE AND ADMINISTRATION OF THE PLAN

2.1 Purpose

The purpose of this Plan is to: (a) aid in attracting, retaining and motivating the directors, officers, employees Directors. Officers, Employees and other eligible Participants of the Salazar Group in the growth and development of the Salazar Group by providing them with the opportunity through RSUs to acquire an increased proprietary interest in the Company; (b) more closely align their interests with those of the Company's shareholders; (c) focus such Participants on operating and financial performance and long-term shareholder value; and (d) motivate and reward for their performance and contributions to the Company's long-term success.

2.2 Administration of the Plan

Subject to Section 2.4, this Plan shall be administered by the Board.

2.3 Authority of the Board

The Board shall have the full power to administer this Plan, including, but not limited to, the authority to:

- (a) interpret and construe any provision hereof and decide all questions of fact arising in their interpretation;
- (b) adopt, amend, suspend and rescind such rules and regulations for administration of this Plan as the Board may deem necessary in order to comply with the requirements of this Plan, or in order to conform to any law or regulation or to any change in any laws or regulations applicable thereto;
- (c) determine the individuals or companies to whom RSUs may be awarded provided that the Board, together with such individuals or companies, are responsible for ensuring and confirming that such person is a bona fide employee, consultant or management company employee Employee, Consultant or Management Company Employee of any member of the Salazar Group and therefore eligible as a Participant;
- (d) award such RSUs on such terms and conditions as it determines including, without limitation: the time or times at which RSUs may be awarded; the time or times when each RSU shall vest and the term of each RSU; whether restrictions or limitations are to be imposed on the Shares the Company may elect to issue in settlement of all or a portion of the Award Value of vested RSUs and the nature of such restrictions or limitations, if any; any acceleration or waiver of termination or forfeiture regarding any RSU; in each case, based on such factors as the Board may determine appropriate, in its sole discretion;
- (e) take any and all actions permitted by this Plan; and
- (f) make any other determinations and take such other action in connection with the administration of this Plan that it deems necessary or advisable.

2.4 Delegation of Authority

To the extent permitted by applicable law, the Board may, from time to time, delegate to a committee (the "Committee") of the Board all or any of the powers conferred on the Board under this Plan. In such event, the Committee will exercise the powers delegated to it by the Board in the manner and on the terms authorized by the Board. Any decision made or action taken by the Committee arising out of or in connection with the administration or interpretation of this Plan in this context is final and conclusive.

The Board or the Committee may delegate or sub-delegate to any director or officer of the Company the whole or any part of the administration of this Plan and shall determine the scope of such delegation or sub-delegation in its sole discretion.

2.5 Discretionary Relief

Notwithstanding any other provision hereof, the Board may, in its sole discretion, waive any condition set out herein if it determines that specific individual circumstances warrant such waiver.

2.6 Amendment or Discontinuance of the Plan

- The Board may amend this Plan in any way, or discontinue this Plan altogether, and may (a) amend, in any way, any RSU granted under this Plan at any time without the consent of a Participant, provided that such amendment shall not adversely alter or impair any RSU previously granted under the Plan or any related RSU Agreement, except as otherwise permitted hereunder and further provided that no amendment will cause the Plan or any RSU to cease to comply with paragraph (k) of the definition of "salary deferral arrangement" in subsection 248(1) of the Income Tax Act (Canada). In addition, the Board may, by resolution, make any amendment to this Plan or any RSU granted under it (together with any related RSU Agreement) without shareholder approval, provided however, that the Board will not be entitled to amend this Plan or any RSU granted under it without shareholder (disinterested shareholder approval Disinterested Shareholder Approval if applicable) and, if applicable, TSXV approval, in order to: (i) increase the maximum number of Shares issuable pursuant to this Plan; (ii) cancel an RSU and subsequently issue to the holder of such RSU a new RSU in replacement thereof; (iii) extend the term of an RSU, but not beyond the Expiry Date; (iv) permit the assignment or transfer of an RSU other than as provided for in this Plan; (v) add to the categories of persons eligible to participate in this Plan; (vi) remove or amend Sections 4.4 (b), (c), (d) or (e) of this Plan; (vii) remove or amend this Section 2.6(a); or (viii) in any other circumstances where TSXV and shareholder approval is required by the TSXV. Any renewal of this plan will be subject to disinterested shareholder approval Disinterested Shareholder Approval, and TSXV approval as applicable.
- (b) <u>Disinterested Shareholder Approval is required for any extension of the term of an RSU, if</u> the RSU holder is an Insider of the Company at the time of the proposed amendment.
- (c) (b)-Without limitation of Section 2.6(a), the Board may correct any defect or supply any omission or reconcile any inconsistency in this Plan in the manner and to the extent deemed necessary or desirable, may establish, amend, and rescind any rules and regulations relating to this Plan, and may make such determinations as it deems necessary or desirable for the administration of this Plan.
- (d) (e) On termination of this Plan, any outstanding awards of RSUs under this Plan shall immediately vest and the Award Value underlying the RSUs shall be paid to the Participants in accordance with and upon compliance with Section 4.6. This Plan will finally cease to operate for all purposes when (i) the last remaining Participant receives payment in respect of the Award Value underlying all RSUs credited to the Participant's Account, or (ii) all unvested RSUs expire in accordance with the terms of this Plan and the relevant RSU Agreements.

2.7 Final Determination

Any determination or decision by, or opinion of, the Board, the Committee or a director or officer of the Company made or held pursuant to the terms set out herein shall be made or held reasonably and shall be final, conclusive and binding on all parties concerned, including, but not limited to, the Company, the Participants and their beneficiaries and legal representatives.

Subject to Section 2.5, all rights, entitlements and obligations of Participants under this Plan are set forth in the terms hereof and cannot be modified by any other documents, statements or communications, except by amendment to the terms set out herein referred to in Section 2.6.

2.8 Withholding Taxes

When a Participant or other person becomes entitled to receive a payment in respect of any RSUs, the Company or a member of the Salazar Group shall have the right to require the Participant or such other person to remit to the Company or to a member of the Salazar Group, as the case may be, an amount sufficient to satisfy any withholding tax requirements relating thereto. Unless otherwise prohibited by the Committee or by applicable law, satisfaction of the withholding tax obligation may be accomplished by any of the following methods or by a combination of such methods:

- (a) the tendering by the Participant of a cash payment to the Company, or a member of the Salazar Group, as the case may be;
- (b) where the Company has elected to issue Shares to the Participant, the withholding by the Company or a member of the Salazar Group, as the case may be, from the Shares otherwise deliverable to the Participant such number of Shares as it determines are required to be sold by the Company, or a member of the Salazar Group, as the case may be, as agent for and on behalf of the Participant, to satisfy the total withholding tax obligation (net of selling costs, which shall be paid by the Participant). The Participant consents to such sale and grants to the Company, or a member of the Salazar Group, as the case may be, an irrevocable power of attorney to effect the sale of such Shares and acknowledges and agrees that neither the Company nor any member of the Salazar Group accepts any responsibility for the price obtained on the sale of such Shares; or
- (c) the withholding by the Company or a member of the Salazar Group, as the case may be, from any cash payment otherwise due to the Participant;

provided, however, that the sum of any cash so paid or withheld and the Fair Market Value of any Shares so withheld is sufficient to satisfy the total withholding tax obligation. Any reference in this Plan to the Award Value or payment of cash or issuance of Shares in settlement thereof is expressly subject to this Section 2.8.

2.9 Taxes

Participants (or their beneficiaries) shall be responsible for reporting and paying all taxes with respect to any RSUs under the Plan, whether arising as a result of the grant or vesting of RSUs or otherwise. Neither the Company nor the Board make any guarantees to any person regarding the tax treatment of an RSU or payments made under the Plan and none of the Company or any of its employees or representatives shall have any liability to a Participant with respect thereto. In the event the Company applies in a local jurisdiction for a favourable and/or reduced tax route in such jurisdiction, and if the Plan or the grant fails to qualify for this reduced tax route, for any reason, the Participant in this jurisdiction shall bear the full responsibility for the taxes and the Company shall bear no liability what so ever to the Participant for such tax treatment. The Company will provide each Participant with (or cause each Participant to be provided with) a T4 slip or such information return as may be required by applicable law to report income, if any, arising upon the grant or vesting of rights under this Plan by a Participant for income tax purposes.

2.10 Information

Each Participant shall provide the Company with all of the information (including personal information) that it requires in order to administer this Plan.

2.11 Account Information

Information pertaining to the RSUs in Participants' Accounts will be made available to the Participants at least annually in such manner as the Company may determine and shall include such matters as the Board or the Committee may determine from time to time or as otherwise may be required by law.

2.12 Indemnification

Each member of the Board or Committee is indemnified and held harmless by the Company against any cost or expense (including any sum paid in settlement of a claim with the approval of the Company) arising out of any act or omission to act in connection with the terms hereof to the extent permitted by applicable law. This indemnification is in addition to any rights of indemnification a Board or Committee member may have as director or otherwise under the Articles of the Company, any agreement, any vote of shareholders, or disinterested directors, or otherwise.

ARTICLE III ELIGIBILITY AND PARTICIPATION IN THE PLAN

3.1 Participation

The Board, in its sole discretion, shall determine, or shall delegate to the Committee the authority to determine, which Participants will participate in this Plan.

3.2 RSU Agreement

A Participant shall confirm acknowledgement of an award of RSUs made to such Participant in such form as determined by the Board from time to time (the "RSU Agreement"), within such time period and in such manner as specified by the Board. If acknowledgement of an award of RSUs is not confirmed by a Participant within the time specified, the Company reserves the right to revoke the crediting of RSUs to the Participant's Account.

3.3 Participant's Agreement to be Bound

Participation in this Plan by any Participant shall be construed as irrevocable acceptance by the Participant of the terms and conditions set out herein and all rules and procedures adopted hereunder and as amended from time to time.

ARTICLE IV TERMS OF THE PLAN

4.1 Grant of RSUs

Subject to Section 3.2, an award of RSUs pursuant to this Plan will be made and the number of such RSUs awarded will be credited to each Participant's Account, effective as of the Award Date. The number of RSUs to be credited to each Participant's Account shall be determined by the Board, or the Committee delegated by the Board to do so, each in its sole discretion.

Investor Relations Service Providers may not receive any RSUs pursuant to this Plan.

4.2 Credits for Dividends

Following the declaration and payment of dividends on the Shares, the Board may, in its absolute discretion, determine to make a cash payment to a Participant in respect of outstanding vested RSUs credited to the Participant's Account and subject to the Participant providing an executed exercise notice (a "**Dividend Equivalent**"). Such Dividend Equivalent, if any, shall be computed by dividing: (a) the amount obtained by multiplying the amount of the dividend declared and paid per Share by the number of RSUs recorded in the Participant's Account on the record date for the payment of such dividend, by (b) the Dividend Market Value, with fractions computed to three decimal places. Payment of any such Dividend Equivalent will be made forthwith following any such determination by the Board and in any event within thirty (30) days of such determination.

4.3 Vesting

The Board or the Committee may, in its sole discretion, determine the time during which RSUs shall vest (except that no RSU, or portion thereof, may vest after the Expiry Date) and whether there shall be any other conditions or performance criteria to vesting. Notwithstanding the foregoing, the Committee may, at its sole discretion at any time or in the RSU Agreement in respect of any RSUs granted, accelerate or provide for the acceleration of vesting in whole or in part of RSUs previously granted. The Award Value of any RSU shall be determined as of the applicable Vesting Date.

4.4 Limits on Issuances

Notwithstanding any other provision of this Plan:

- (a) the maximum number of Shares issuable at any time pursuant to RSUs under this Plan shall be limited to 1,000,000, subject to the number of Shares issuable at any time under all Security Based Compensation Arrangements under all Security Based Compensation Arrangements (including, for greater certainty, pursuant to RSUs issued under this Plan) not exceeding 10% of the issued and outstanding Shares;
- (b) the number of Shares reserved for issuance to Participants retained to provide Investor Relations Activities under all Security Based Compensation Arrangements in any 12 month period shall not exceed 2% of the issued and outstanding Shares;

- (b) the <u>maximum aggregate</u> number of Shares reserved for issuance to any one Participant retained as a <u>consultantConsultant</u> to provide services to any of the entities comprising the Salazar Group under all Security Based Compensation Arrangements in any 12 month period shall not exceed 2% of the issued and outstanding Shares <u>calculated</u> as at the date of grant or issue to the Consultant:
- (d) unless the Company has received disinterested shareholder approval Disinterested Shareholder Approval to do so, the number of Shares reserved for issuance to any one Participant under all Security Based Compensation Arrangements, including the Company's stock option plan, in any 12 month period will not exceed 5% of the issued and outstanding Shares calculated as at the date of the grant to such Participant;
- (d) (e) unless the Company has received disinterested shareholder approval Disinterested Shareholder Approval to do so the maximum aggregate number of Shares issuable to Insiders (as a group), at any time, under all Security Based Compensation Arrangements, including the Company's stock option plan, shall not exceed 10% of the issued and outstanding Shares at any point in time; and
- (e) (f)-unless the Company has received disinterested shareholder approval Disinterested Shareholder Approval to do so the number of Shares issued to Insiders (as a group), within any one year period, under all Security Based Compensation Arrangements, including the Company's stock option plan, shall not exceed 10% of the issued and outstanding Shares calculated at the date of the grant to any Insider.

For the purposes of this Section 4.4, any increase in the issued and outstanding Shares (whether as a result of the issue of Shares from treasury in settlement of the Award Value underlying vested RSUs or otherwise) will not increase the number of Shares that may be issued pursuant to this Plan. Shares issued from treasury in settlement of an Award Value underlying vested RSUs will not become available for grant under this Plan.

RSUs (or the Award Value thereof) that are cancelled, surrendered, terminated or that expire prior to the final Vesting Date or in respect of which the Company has not elected to issue Shares from treasury in respect thereof shall result in such Shares that were reserved for issuance thereunder being available to be issued, at the election of Company, in respect of a subsequent grant of RSUs pursuant to this Plan to the extent of any Shares which have not been issued from treasury in respect of any such RSU.

For purposes of the calculations in this Section 4.4 only, it shall be assumed that all issued and outstanding RSUs will be settled by the issuance of Shares from treasury, notwithstanding the Company's right pursuant to Section 4.6 to settle the Award Value underlying vested RSUs in cash (if applicable) or by purchasing Shares on the open market.

In addition to the terms set out herein, the administration and limitations of this Plan will be subject to the provisions of TSXV Policy 4.4 – *Incentive Stock Options*, as applicable.

4.5 RSU Terms

The term during which a RSU may be outstanding shall, subject to the provisions of this Plan requiring or permitting the acceleration or the extension of the term, be such period as may be determined from time to time by the Board or the Committee, but subject to the rules of any stock exchange or other regulatory body having jurisdiction (but in no case shall the term of an RSU extend beyond the Expiry Date).

In addition, unless otherwise determined by the Board or the Committee, or unless the Company and a Participant agree otherwise in an RSU Agreement or other written agreement (including an employment or consulting agreement), each RSU shall provide that if a Participant shall cease to be a director Director or officer Officer of or be in the employ of, or a consultant Consultant or other Participant to, any of the entities comprising the Salazar Group for any reason whatsoever including, without limitation, retirement, resignation or involuntary termination (with or without cause), as determined by the Board in its sole discretion, before all of the awards respecting RSUs credited to the Participant's Account have vested or are forfeited pursuant to any other provision hereof, (i) such Participant shall cease to be a Participant as of the Forfeiture Date, (ii) the former Participant shall forfeit all unvested awards respecting RSUs credited to the Participant's Account effective as at the Forfeiture Date, (iii) any Award Value corresponding to any vested RSUs remaining unpaid as of the Forfeiture Date shall be paid to the former Participant in accordance with Section 4.6, and (iv) the former Participant shall not be entitled to any further payment from this Plan.

Notwithstanding the preceding paragraph or anything else contained in this Plan to the contrary, unless otherwise determined by the Board or the Committee, or unless the Company and a Participant agree otherwise in an RSU Agreement or other written agreement (including an employment or consulting agreement), if a Participant shall cease to be a director Director or officer Officer of or be in the employ of, or a consultant Consultant or other Participant to, any of the entities comprising the Salazar Group due to the death of the Participant, any unvested RSUs in the deceased Participant's Account effective as at the time of the Participant's death shall be deemed to have vested immediately prior to the Forfeiture Date with the result that the deceased Participant shall not forfeit any unvested RSUs and the Award Value corresponding to all RSUs credited to such Participant's Account shall be paid to the legal representative of the deceased former Participant's estate in accordance with Section 4.6 after receipt of satisfactory evidence of the Participant's death from the authorized legal representative of the deceased Participant.

Where a Vesting Date occurs on a date when a Participant is subject to a Black-Out Period, such Vesting Date shall be extended to a date which is within (10) ten business days following the end of such Black-Out Period, and further provided that (i) if any such extension would cause the Vesting Date or Vesting Dates to extend beyond the Expiry Date, the amounts to be paid on such Vesting Date or Vesting Dates shall be paid on the Expiry Date notwithstanding the Black-Out Period, and (ii) if a Forfeiture Date occurs in respect of a Participant after the original Vesting Date then any unvested RSUs credited to the Participant's Account effective as of the Forfeiture Date that would have vested as of the original Vesting Date but for the Black-Out Period, shall be deemed to have vested immediately prior to the Forfeiture Date, but, subject to subparagraph (i), the Award Value of any such-vested RSUs shall be determined as of the Vesting Date as so extended by the provisions above, and any payment thereof shall be made only after such determination.

The automatic extension of a Vesting Date pursuant to this section 4.5 is available to all eligible RSU holders under this Plan under the same terms and conditions, provided however, that such automatic extension will not be permitted where the RSU holder or the Company is subject to a cease trade order (or similar order under Securities Laws) in respect of the Company's securities.

This Plan does not confer upon a Participant any right with respect to continuation of employment by or service provision to any of the entities comprising the Salazar Group, nor does it interfere in any way with the right of the Participant or any of the entities comprising the Salazar Group to terminate the Participant's employment or service provision at any time.

4.6 Payment in Respect of RSUs

On the Vesting Date, the Company, at its sole and absolute discretion, shall have the option of settling the Award Value payable in respect of an RSU by any of the following methods or by a combination of such methods:

- (a) payment in cash;
- (b) payment in Shares acquired by the Company on the Exchange; or
- (c) payment in Shares issued from the treasury of the Company.

The Company shall not determine whether the payment method shall take the form of cash (if applicable) or Shares until the Vesting Date, or some reasonable time prior thereto. A holder of RSUs shall not have any right to demand, be paid in, or receive Shares in respect of the Award Value underlying any RSU at any time. Notwithstanding any election by the Company to settle the Award Value of any vested RSUs, or portion thereof, in Shares, the Company reserves the right to change its election in respect thereof at any time up until payment is actually made, and the holder of such vested RSUs shall not have the right, at any time to enforce settlement in the form of Shares of the Company.

Any amount payable to a Participant in respect of vested RSUs shall be paid to the Participant as soon as practicable following the Vesting Date and in any event within thirty (30) days of the Vesting Date and prior to the Outside Payment Date (provided that any amount payable with respect to a Vesting Date that occurs after the Forfeiture, but before the RSU has terminated in accordance with an applicable provision of Section 4.6, must occur not later than the Expiry Date).

Where the Company elects to pay any amounts pursuant to vested RSUs by issuing Shares, and the determination of the number of Shares to be delivered to a Participant in respect of a particular Vesting Date would result in the issuance of a fractional Share, the number of Shares deliverable on the Vesting Date shall be rounded down to the next whole number of Shares. No certificates representing fractional Shares shall be delivered pursuant to this Plan nor shall any cash amount be paid at any time in lieu of any such fractional interest.

ARTICLE V EFFECT OF CORPORATE EVENTS

5.1 Alterations in Shares

In the event:

- (a) of any change in the Shares through subdivision, consolidation, reclassification, amalgamation, merger or otherwise; or
- (b) that any rights are granted to all or substantially all shareholders to purchase Shares at prices substantially below Fair Market Value; or
- (c) that, as a result of any recapitalization, merger, consolidation or other transaction, the Shares are converted into or exchangeable for any other securities or property;

then the Board may make such adjustments to this Plan, to any RSUs and to any RSU Agreements outstanding under this Plan as the Board may, in its sole discretion, consider appropriate in the circumstances to prevent dilution or enlargement of amounts to be paid to Participants hereunder.

For greater certainty, any adjustment, other than in connection with a share consolidation or share split, to RSUs granted or issued under this Plan are subject to the prior acceptance of the

Exchange, including adjustments related to an amalgamation, merger, arrangement, reorganization, spin-off, dividend or recapitalization.

5.2 Merger and Sale, etc.

Except in the case of a transaction that is a Change of Control and to which Section 5.3 applies. if the Company enters into any transaction or series of transactions whereby the Company or all or substantially all of the assets would become the property of any other trust, body corporate, partnership or other person (a "Successor"), whether by way of takeover bid, acquisition, reorganization, consolidation, amalgamation, arrangement, merger, transfer, sale or otherwise, prior to or contemporaneously with the consummation of such transaction the Company and the Successor will execute such instruments and do such things as the Board or the Committee may determine are necessary to establish that upon the consummation of such transaction the Successor will assume the covenants and obligations of the Company under this Plan and the RSU Agreements outstanding on consummation of such transaction. Any such Successor shall succeed to, and be substituted for, and may exercise every right and power of the Company under this Plan and RSU Agreements with the same effect as though the Successor had been named as the Company herein and therein and thereafter, the Company shall be relieved of all obligations and covenants under this Plan and such RSU Agreements and the obligation of the Company to the Participants in respect of the RSUs shall terminate and be at an end and the Participants shall cease to have any further rights in respect thereof including, without limitation, any right to acquire Shares upon vesting of the RSUs.

5.3 Change of Control

Notwithstanding any other provision in this Plan but subject to any provision to the contrary contained in an RSU Agreement or other written agreement (such as an agreement of employment) between the Company and a Participant, if there takes place a Change of Control, the Board or the Committee shall have the absolute discretion to determine if all issued and outstanding RSUs shall vest (whether or not then vested) upon the Change of Control and the Vesting Date shall be the date which is immediately prior to the time such Change of Control takes place, or at such earlier time as may be established by the Board or the Committee, in its absolute discretion, prior to the time such Change of Control takes place.

ARTICLE VI GENERAL

6.1 Compliance with Laws

The Company, in its sole discretion, may postpone the issuance or delivery of any Shares that it elects to issue pursuant to any RSU to such date as the Committee may consider appropriate, and may require any Participant to make such representations and furnish such information as it may consider appropriate in connection with the issuance or delivery of Shares in compliance with applicable laws, rules and regulations, except that in no event may the issuance of such Shares in respect of a RSU occur after the Outside Payment Date. The Company shall not be required to qualify for resale pursuant to a prospectus or similar document any Shares that it elects to issue pursuant to the Plan, provided that, if required, the Company shall notify the Exchange and any other appropriate regulatory bodies in Canada and the United States of the existence of the Plan and the granting of RSUs hereunder in accordance with any such requirements.

6.2 General Restrictions and Assignment

Except as required by law, the rights of a Participant hereunder are not capable of being assigned, transferred, alienated, sold, encumbered, pledged, mortgaged or charged and are not

capable of being subject to attachment or legal process for the payment of any debts or obligations of the Participant.

The rights and obligations hereunder may be assigned by the Company to a Successor to the business of the Company.

6.3 Market Fluctuations

No amount will be paid to, or in respect of, a Participant under this Plan to compensate for a downward fluctuation in the price of Shares, nor will any other form of benefit be conferred upon, or in respect of, a Participant for such purpose. The Plan will be unfunded.

The Company makes no representations or warranties to Participants with respect to this Plan or the RSUs whatsoever. Participants are expressly advised that the value of any RSUs and Shares under this Plan will fluctuate as the trading price of Shares fluctuates.

In seeking the benefits of participation in this Plan, a Participant agrees to exclusively accept all risks associated with a decline in the market price of Shares and all other risks associated with the holding of RSUs.

6.4 No Shareholder Rights

Until Shares have actually been issued and delivered should the Company elect to so issue Shares in accordance with the terms of the Plan, a Participant to whom RSUs have been granted shall not possess any incidents of ownership of such Shares including, for greater certainty and without limitation, the right to receive dividends, if any, on such Shares and the right to exercise voting rights in respect of such Shares.

6.5 Section 409A

This Plan, the RSUs and payments made to U.S. Participants pursuant to this Plan are intended to comply with, or qualify for an exemption from, the requirements of Section 409A of the Code and shall be construed consistently therewith and shall be interpreted in a manner consistent with that intention. Terms defined in this Plan shall have the meanings given to such terms under Section 409A of the Code if and to the extent required to comply with Section 409A. Notwithstanding any other provision of this Plan, the Company reserves the right, to the extent it deems necessary or advisable, in its sole discretion, to unilaterally amend the Plan to ensure that all RSUs issued to U.S. Participants are awarded in a manner that qualifies for exemption from, or complies with, Section 409A, provided, however, that the Company makes no undertaking to preclude Section 409A from applying to an award of RSUs, and the U.S. Participant or his or her estate, as the case may be, is and shall at all times be solely responsible for the payment of all taxes and penalties under Section 409A. The Company, its affiliates, directors, officers and agents shall have no liability to a U.S. Participant, or any other party, if an RSU that is intended to be exempt from, or compliant with, Section 409A is not so exempt or compliant, or for any action taken by the Committee.

6.6 Governing Law

The validity, construction and effect of this Plan and any actions taken or relating to this Plan shall be governed by the laws of the Province of British Columbia and the federal laws of Canada applicable therein.

6.7 Currency

All amounts paid or values to be determined under this Plan shall be in Canadian dollars.

6.8 Severability

The invalidity or unenforceability of any provision of this document shall not affect the validity or enforceability of any other provision and any invalid or unenforceable provision shall be severed from this document.

6.9 Effective Time

This Plan shall be effective as of ◆ September 25, 2020, as amended December 6, 2022.

54510386\1

54510386\2